UNITED STATES BANKRUPTCY COURT EASTERN DISTRICT OF NORTH CAROLINA WILSON DIVISION

In re:)
NATIONAL GAS DISTRIBUTORS, LLC,) Case No. 06-00166-8-ATS) Chapter 11
Debtor.))
RICHARD M. HUTSON, II, TRUSTEE FOR NATIONAL AS DISTRIBUTORS, LLC,) Adversary Proc.) No. 06-00267-8-ATS
f/k/a Paul Lawing, Jr., LLC,))
Plaintiff,))
v.	,))
THE SMITHFIELD PACKING COMPANY, INCORPORATED,)))
Defendant.	ý))

MEMORANDUM IN SUPPORT OF DEFENDANT'S MOTION: (A) TO DISMISS TRUSTEE'S COMPLAINT FOR FAILURE TO STATE A CLAIM UNDER FEDERAL RULE OF BANKRUPTCY PROCEDURE 7012, OR, IN THE ALTERNATIVE, (B) FOR SUMMARY JUDGMENT UNDER FEDERAL RULE OF BANKRUPTCY PROCEDURE 7056

NOW COMES Defendant The Smithfield Packing Company, Incorporated ("Defendant"), by and through counsel, and files this Memorandum (the "Supporting Memorandum") in Support of Defendant's Motion: (A) To Dismiss The Trustee's Complaint For Failure To State A Claim Under Federal Rule Of Bankruptcy Procedure 7012, or, in The Alternative, (B) For Summary Judgment Under Federal Rule Of Bankruptcy Procedure 7056 (the "Motion"). In support of the Motion, Defendant respectfully states as follows:

I. INTRODUCTION AND BACKGROUND

A "derivative" is simply "an agreement between a future buyer and future seller." Michael Durbin, <u>All About Derivatives</u> (McGraw-Hill 2006), p. 1. In addition to specifying the buyer and the seller, every derivative specifies a future price at which some item can or must be sold and a future date on or before which such sale must occur. <u>Id.</u>

A "forward contract" is the simplest type of derivative – a forward contract obligates one party to buy and the other party to sell a particular item at a set price on some date in the future. Id. at 16.¹ In a forward contract relating to a commodity such as natural gas, the price of which fluctuates and cannot be predicted into the future with certainty, one virtual certainty does exist – one party to the contract, either buyer or seller, will have made a "bad bet." The parties will not know the "loser" on the bet or the magnitude of the loser's loss (or, conversely, the winner's gain) until delivery occurs under the forward contract. At that point in time, but not before, the parties will be able to judge whether the forward contract price is higher or lower than the market price of the commodity in question at the time of delivery.

Derivatives such as forward contracts have become a mainstay of the worldwide economy. As of the middle of 2006, there was a notional amount outstanding of interest rate swaps, options and currency swaps, and equity derivatives in excess of \$250 trillion. See International Swaps and Derivatives Association ("ISDA") 2006 Mid-Year

_

¹ A forward contract's simplicity distinguishes it from certain other types of derivatives, such as a "futures contract" (a standardized forward contract executed at and tradable on an exchange) or an "option contract" (which grants the holder the right but not the obligation to buy or sell something at a specified price on or before a specified future date). Michael Durbin, <u>All About Derivatives</u> (McGraw-Hill 2006), p. 2.

² The theoretical possibility exists, of course, that the contract price and the actual market price of the

² The theoretical possibility exists, of course, that the contract price and the actual market price of the commodity at the time of delivery will be precisely the same. The likelihood of this is null when the forward contract, as in this case, involves daily deliveries of a volatile commodity such as natural gas at the contract price over a multi-month period.

Summary, available at www.isda.org/statistics. Derivative contracts, such as "swap agreements," "forward contracts," "forward agreements," "options," and other "future" agreements, especially in the sale of commodities, are different from standard sale or supply contracts because such derivative contracts are risk-shifting in nature and their purpose is to hedge against fluctuations in the commodities market. H.R. REP. No. 101-484, at 4 (1990), reprinted in 1990 U.S.C.C.A.N. 223, 226. Thus, such derivative contracts have long been accorded special treatment under the United States Bankruptcy Code, 11 U.S.C. § 101, et seq. (the "Code"), to avoid the ripple effect that could lead to the collapse of an affected market. Accordingly, the Code provides substantive protections and exemptions to parties to derivative contracts, excepting setoffs under such contracts from the effect of the automatic stay, excepting such contracts from the general unenforceability of ipso facto clauses, and, most pertinently, limiting the ability of a trustee to avoid, as preferences or fraudulent transfers, transfers made in connection with such contracts. See, e.g., 11 U.S.C. §§ 546(e)-(g), 362(b)(6), and 362(b)(7).

Among the effects of these exemptive provisions in the Code is that the "winner" of the bet in a forward transaction is not penalized with fraudulent transfer liability if the "loser" becomes a debtor in bankruptcy shortly after the transaction. Such transactions can **never** result in constructive fraudulent transfer liability under the Code for the winner of the bet, no matter how big a "win" the winner enjoys. Even if the debtor entered into the forward transaction with actual fraudulent intent, such transaction is exempt from avoidance provided that (with respect to actual fraud only) the "winner" entered into the transaction in good faith.

In this case, the Trustee has ignored the fact that the "Transfers" (hereinafter, as

defined in the Complaint³) that he seeks to avoid were made under and in connection with "swap agreements" and he has filed the Complaint seeking avoidance and recovery of the Transfers in violation of the specific provisions of the Code that protect these transactions.

The Transfers that the Trustee seeks to avoid in this Complaint relate to natural gas delivered under forward contracts by National Gas Distributors, LLC, the debtor herein (the "Debtor"), to Defendant in the year preceding the Debtor's bankruptcy petition date of January 20, 2006. The terms and conditions of the forward contracts were set forth in a form "Base Contract for Sale and Purchase of Natural Gas" promulgated by the North American Energy Standards Board, Inc. ("NAESB") (the "Base Contract"), a copy of which was attached to the proof of claim (the "Proof of Claim") filed by Defendant in the bankruptcy case. (A copy of the Proof of Claim is attached hereto as Exhibit A.) Generally, the Debtor would confirm the purchase and sale transaction by an email sent to Defendant setting forth the purchase price for the natural gas for a particular future "Delivery Period" (as defined in the Base Contract). Attached to the Proof of Claim is an email exchange dated August 12, 2005, which effectuated and confirmed the sale of natural gas for the period of November, 2005 through March, 2006 at a capped price of \$6.60 per dekatherm. Also attached hereto as Exhibit B-2 and Exhibit B-3 to the Authentication Declaration of Robert E. Miller, attached hereto as Exhibit B, are email exchanges dated April 2, 2004 and February 25, 2005, which effectuated and confirmed the sale of natural gas, respectively, for the period of April, 2004 through March, 2005 at a capped price of \$5.70 per dekatherm, plus basis

_

³ The use herein of the term "Transfers" as that term is defined in the Complaint is not an admission by Defendant as to any of the allegations in the Complaint.

(gas transport costs), and for the period of April, 2005 through October, 2005, at a capped price of \$5.75 per dekatherm, plus basis. (Hereinafter, the "Base Contract," the April 2, 2004 email, the February 25, 2005 email, the August 12, 2005 email, and any other confirming emails or other oral or written confirmation are referred to collectively as the "Forward Contracts." ⁴)

In his Complaint, the Plaintiff, Richard M. Hutson, II, Chapter 11 Trustee (the "Trustee"), seeks to avoid the Transfers of natural gas from Debtor to Defendant as constructive fraudulent transfers under section 548(a)(1)(B) of the Code, alleging that the Debtor sold such natural gas to Defendant at below-market prices, such that the Debtor received less than reasonably equivalent value on account of such Transfers, and that the Debtor was insolvent at the time of, or rendered insolvent by, the Transfers.

Alternatively, the Trustee alleges that the Debtors made the Transfers with the actual intent to hinder, delay or defraud creditors of the Debtor such that the Transfers are actual fraudulent transfers under section 548(a)(1)(A) of the Code.⁵

The Forward Contracts between the Debtor and Defendant constitute "swap agreements," as that term is defined under section 101(53B) of the Code. Pursuant to

-

⁴ Under the Base Contract between the Debtor and Defendant, the term "Contract" is defined as the "legally-binding relationship established by (I) the Base Contract, (II) any and all binding Transaction Confirmations, and (III) where the parties have selected the Oral Transaction Procedure in Section 1.2 of the Base Contract, any and all transactions that the parties have entered into through an EDI transmission or by telephone, but that have not been confirmed in a binding Transaction Confirmation." The Debtor and Defendant selected the Oral Transaction Procedure referenced in Section 1.2 of the Base Contract. ⁵ In the First Claim of his Complaint, the Trustee apparently seeks to avoid the Forward Contracts themselves or the obligations incurred thereunder as both constructive and actual fraudulent transfers. See Complaint, ¶ 16 ("the contract(s) and any obligation incurred by the Debtor to provide natural gas to Plaintiff [sic]...is [sic] avoidable"). However, because of the broad definition in the Code of "swap agreement" (see pp. 10-13, infra), the forward transactions themselves constitute "swap agreements" triggering the application of sections 546(g) and 548(d)(2)(D). Thus, even if the Forward Contracts were themselves avoided, which would be a gross frustration of Congress' intent to exempt such contracts in their entirety from the incidents of bankruptcy (infra), the transactions themselves fall within the exemptive provisions of the Code. Therefore, the Court should dismiss any claim of the Trustee for the avoidance of the Forward Contracts themselves for the reasons set forth herein.

section 546(g) of the Code, the Transfers are excepted from avoidance as constructive fraudulent transfers because they were made to a swap participant under or in connection with a swap agreement. Accordingly, the Court should dismiss under section 546(g) the Trustee's claims in the Complaint asserted under section 548(a)(1)(B).

Furthermore, section 548(c) of the Code prevents the Trustee's avoidance of the Transfers (or value thereof) as actual fraudulent transfers under section 548(a)(1)(A) of the Code. Section 548(c) prevents the avoidance of the Transfers because the Transfers were made to a swap participant under or in connection with swap agreements, which, under section 548(d)(2)(D), establishes conclusively that Defendant provided value to the extent of such Transfers. Further, though the Trustee in his Complaint acknowledges Defendant's defense under section 548(c) in his Complaint, he fails to allege, and cannot allege, that Defendant failed to receive the Transfers in good faith. Therefore, the Court should dismiss under sections 548(c) and (d)(2)(D) the Trustee's claims in the Complaint for avoidance of actual fraudulent transfers asserted under section 548(a)(1)(A) of the Code.

II. STANDARD FOR DISMISSAL

"The purpose of Rule 12(b)(6) is to test the legal sufficiency of a complaint" and not to "resolve contests surrounding the facts, the merits of a claim, or the applicability of defenses." Presley v. City of Charlottesville, 464 F.3d 480, 483 (4th Cir. 2006) (citing Edwards v. City of Goldsboro, 178 F.3d 231, 243-44 (4th Cir. 1999)). A Rule 12(b)(6) motion to dismiss should be granted "if, after accepting all well-pleaded allegations in the plaintiff's complaint as true, it appears certain that the plaintiff cannot prove any set of facts in support of his claim entitling him to relief." Migdal v. Rowe Price-Fleming Int'l,

_

⁶ <u>See</u> Complaint, ¶ 26.

<u>Inc.</u>, 248 F.3d 321, 325 (4th Cir. 2001); see also <u>Venkatraman v. REI Sys., Inc.</u>, 417 F.3d 418, 420 (4th Cir. 2005).

The Court should dismiss the Trustee's claims because the Transfers at issue were made under or in connection with "swap agreements," as defined under the Code, and cannot be avoided and recovered from Defendant as fraudulent transfers under the plain language of the Code. Since the Trustee references the Proof of Claim in the Complaint (see Complaint, ¶15), the Court can and should consider, as well as take judicial notice of, the Proof of Claim and the Forward Contracts attached thereto, in determining this Motion. Enron Corp. v. Citigroup Inc., 2004 Bankr. Lex. 2124, *14 (Bankr. S.D.N.Y. 2004)("In reviewing a 12(b)(6) motion, the court may consider the allegations in the complaint, exhibits attached to the complaint or incorporated therein by reference, matters of which judicial notice may be taken, and documents of which plaintiff has notice and on which it relied in bringing its claim or that are integral to its claim. . . . As such, the document relied upon in framing the complaint is considered to be merged into the pleading.")(internal citation omitted).

Alternatively, Defendant requests the Court determine the Motion under Rule 56 of the Federal Rules of Civil Procedure, made applicable to this matter by Bankruptcy Rules 9014 and 7056. Under Rule 56, a party is entitled to summary judgment when there is no genuine issue of material fact, and the facts, viewed in the light most favorable to the non-moving party, entitle the moving party to judgment in its favor as a matter of law. Hagan v. McNallen (In re McNallen), 62 F.3d 619, 623 (4th Cir. 1995). Summary judgment is not a "disfavored procedural shortcut," but an important mechanism for filtering out "claims and defenses that have no factual basis." Celotex Corp. v. Catrett,

477 U.S. 317, 327 (1986). The Court should grant summary judgment if there exists no genuine issue as to any material fact and the moving party is entitled to judgment as a matter of law. Scott v. Montgomery County Gov't, 164 F. Supp. 2d 502, 505 (D. Md. 2001). "A party who bears the burden of proof on a particular claim must factually support each element of his or her claim." Id. "[A] complete failure of proof concerning an essential element . . . necessarily renders all other facts immaterial." Celotex Corp., 477 U.S. at 323. Significantly, a "mere scintilla of evidence is not enough to create a fact issue." Barwick v. Celotex Corp., 736 F.2d 946, 958-59 (4th Cir. 1984).

As discussed below, it is undisputed that the Debtor made the Transfers under and in connection with the Forward Contracts, that such Forward Contracts and the transactions themselves were "forward contracts" and "swap agreements" under the Code definitions, and that the Transfers are, therefore, not subject to avoidance or recovery from Defendant as fraudulent transfers.

III. ARGUMENT

A. The Transfers Are Not Avoidable As Constructive Fraudulent Transfers
Under Section 548(a)(1)(B) Of The Code Because The Transfers Were Made
To A Swap Participant Under Or In Connection With A Swap Agreement
And Are, Therefore, Excepted From Avoidance As Constructive Fraudulent
Conveyances Under Section 546(g) Of The Code.

Section 546(g) of the Code excepts from avoidance as constructive fraudulent transfers any transfer made under or in connection with a swap agreement:

Notwithstanding section 544, 545, 547, 548(a)(1)(B), and 548(b) of this title, the trustee may not avoid a transfer, made to a swap participant or financial participant, under or in connection with any swap agreement and that is made before the commencement of the case, except under section 548(a)(1)(A) of this title.

11 U.S.C. § 546(g).

Beginning in the original Code in 1978, Congress has included a number of provisions in the Code, including sections 546(e)-(g), 362(b)(6), and 362(b)(7), designed to protect certain transactions in and affecting financial markets. As noted by Congress in the legislative history for the 1990 amendments to the Code:

U.S. bankruptcy law has long accorded special treatment to transactions involving financial markets, to minimize volatility. Because financial markets can change significantly in a matter of days, or even hours, a non-bankrupt party to ongoing securities and other financial transactions could face heavy losses unless the transactions are resolved promptly and with finality.

H.R. REP. No. 101-484, at 2 (1990), reprinted in 1990 U.S.C.C.A.N. 223, 224.

Specifically, "[s]ection 546 of the Bankruptcy Code provides a 'safe harbor' for certain types of transactions. The purpose of section 546 is 'to protect the nation's financial markets from the instability caused by the reversal of [covered] transactions." Enron Corp v. J.P. Morgan Sec., Inc. (In re Enron Corp.), 325 B.R. 671, 684 (Bankr. S.D.N.Y. 2005) (quoting Kaiser Steel Corp. v. Charles Schwab & Co., Inc. (In re Kaiser Steel Corp.), 913 F.2d 846, 848 (10th Cir. 1990)). These safe harbors include section 546(g) as well as analogous provisions under other subsections of 546 protecting other derivative financial transactions. Bankruptcy courts have routinely applied these safe harbor provisions to insulate from avoidance actions financial transactions such as swap agreements, including forward agreements, and related transactions. See e.g., BCP Liquidating LLC v. Bridgeline Gas Mktg., LLC (In re Borden Chems. & Plastics Operating L.P.), 336 B.R. 214 (Bankr. D. Del. 2006)(summary judgment granted for defendant where court found challenged payments were "settlement payments" under "forward contracts" and therefore not avoidable as preferences, under section 546(e)); Enron Corp. v. Credit Suisse First Boston Int'l (In re Enron Corp.), 328 B.R. 58 (Bankr.

S.D.N.Y. 2005), aff'd, 2006 U.S. Dist. LEXIS 57422 (S.D.N.Y. May 2, 2006)(court recognized defense to preference and fraudulent transfer claims if transfers occurred under "swap agreement," pursuant to sections 546(e) and (g)); Williams v. Morgan Stanley Capital Group, Inc. (In re Olympic Natural Gas Co.), 258 B.R. 161 (Bankr. S.D. Tex. 2001), aff'd, 294 F.3d 737 (5th Cir. 2002)(summary judgment granted for defendant where court found that payments were "settlement payments" under "forward contracts" and therefore not avoidable as preferences under section 546(e)); Interbulk, Ltd. v. Louis Dreyfus Corp. (In re Interbulk, Ltd.), 240 B.R. 195, 201 (Bankr. S.D.N.Y. 1999)(in case decided under predecessor version of section 546(g), court recognized defense to preference action for transfers "under a swap agreement").

In 1990, Congress added Section 546(g) to the Code for the purpose of prohibiting:

a bankruptcy trustee from avoiding a transfer under a swap agreement entered into before the bankruptcy petition was filed. An exception is created [from the exemption from avoidance] for any swap agreement entered into with the actual intent to hinder, delay, or defraud any creditor of the debtor, in which case the trustee is permitted to avoid the transfer.

H.R. REP. No. 101-484, at 5 (1990), *reprinted in* 1990 U.S.C.C.A.N. 223, 227. The protections under section 546(g) were clarified and expanded significantly under the October 2005 amendments to the Code under the Bankruptcy Abuse Prevention and Consumer Protection Act of 2005 ("BAPCPA").⁷

_

⁷ One amendment to Section 546(g) under BAPCPA clarified that transfers made under <u>or</u> in connection with a "swap agreement" are not avoidable as constructive fraudulent transfers or preferences. 11 U.S.C.§ 546(g). This amendment clarified the broad intent and application of the statute and settled a statutory ambiguity that had caused one court to find that a transfer must be both "under" and "in connection with" a swap agreement for section 546(g) to apply. <u>See Interbulk, Ltd. v. Louis Dreyfus Corp. (In re Interbulk, Ltd.)</u>, 240 B.R. 195, 201 (Bankr. S.D.N.Y. 1999)(finding that previous version of section 546(g) required transfers be both "under" and "in connection with" swap agreement for section 546(g) to apply). In this case, the Transfers at issue were made **both** under and in connection with "swap agreements."

The BAPCPA amendments substantially broadened the definition of "swap agreement" in the Code. (The full definition of "swap agreement" under section 101(53B) with blacklining showing the changes under BAPCPA is attached hereto as Exhibit C.) As Congress stated in the legislative history regarding this amendment:

As amended, the definition of "swap agreement" will update the statutory definition and achieve contractual netting across economically similar transactions.

The definition of "swap agreement" originally was intended to provide sufficient flexibility to avoid the need to amend the definition as the nature and uses of swap transactions matured. To that end, the phrase "or any other similar agreement" was included in the definition. (The phrase "or any similar agreement" has been added to the definitions of "forward contract," "commodity contract," "repurchase agreement" and "securities contract" for the same reason.)

H.R. REP. No. 109-31, Pt. 1, at 128 (2005), reprinted in Collier on Bankruptcy, App. Pt. 10-403 (2006).⁸

Since the recent amendments, there have been no reported bankruptcy cases that have analyzed or applied the protections under section 546(g) of the Code and the expanded definition of "swap agreement" to a Chapter 5 avoidance action. However, certain commentators noted the significant expanded protection of such transactions, stating:

The expanded definitions--especially the definition of "swap agreement"--are now so broad that **nearly every derivative contract is subject to the Code's protection**. Instead of protecting particular counterparties to particular transactions, **the Code now protects any counterparty to any derivatives contract. Entire markets have been insulated from the costs of a bankruptcy filing by a financial contract counterparty**. Equally important, the amendments limit judicial discretion to assess the economic substance of financial transactions, even those that resemble

_

⁸ As discussed below, the Forward Contracts meet the current definition of "swap agreement." The Forward Contracts would have also satisfied the definition of "swap agreement" before BAPCPA.

ordinary loans or that retire a debtor's outstanding debt or equity. The reforms of 2005 direct judges to apply a formalistic inquiry based on industry custom: a financial transaction is a "swap," "repurchase transaction," or other protected transaction if it is treated as such in the relevant financial market.

. . . .

This shift in the Code effectively eliminates the concept of protected parties with respect to forwards and commodity contracts. Any counterparty to these contracts is a "swap participant" and therefore protected. This conclusion creates no tension with the various provisions--362(b)(6), 546(e), and 556--that permit only certain parties to forward and commodity contracts to enjoy the Code's safe harbors. These provisions protect particular parties, but they do not rule out safe harbors for other counterparties under other provisions of the Code. Indeed, courts have long recognized significant overlap in the Code's definitions, and Congress was fully aware that the new definition of "swap agreement" would cover all forwards. Indeed, legislative history indicates that Congress was aware that all of the Code's definitions overlap considerably.

Edward R. Morrison & Joerg Riegel, <u>Financial Contracts and the New Bankruptcy Code:</u>

<u>Insulating Markets from Bankrupt Debtors and Bankruptcy Judges</u>, 13 Am. Bankr. Inst.

L. Rev. 641, 652 (2005)(emphasis added).

A significant change under BAPCPA to the definition of "swap agreement" was the clarification that "a commodity index or a commodity swap, option, future, or **forward agreement**" falls within the definition of "swap agreement." 11 U.S.C. § 101(53B)(A)(i)(VII)(emphasis added). The inclusion of "forward agreement" in the expanded definition of "swap agreement" was intended to cover not only "forward contracts," as that term is defined in section 101(25) of the Code (see infra), but also any forward transaction. H.R. REP. No. 109-31, Pt. 1, at 129 (2005), *reprinted in* COLLIER ON BANKRUPTCY, App. Pt. 10-405 (2006) ("The use of the term 'forward' in the definition of 'swap agreement' is not intended to refer only to transactions that fall within

the definition of 'forward contract.' Instead, a 'forward' transaction could be a 'swap agreement' even if not a 'forward contract.'"). Thus, even an undocumented forward transaction constitutes a "swap agreement."

i. The Forward Contracts At Issue Are "Forward Agreements" And "Swap Agreements" As Those Terms Are Defined Under The Code And Applicable Law.

Under section 101(25), "forward contract" is defined, to include, in pertinent part:

- (A) a contract (other than a commodity contract) for the purchase, sale, or transfer of a commodity, as defined in section 761(8) of this title, or any similar good, article, service, right, or interest which is presently or in the future becomes the subject of dealing in the forward contract trade, or product or byproduct thereof, with a maturity date more than two days after the date the contract is entered into, including, but not limited to, a repurchase transaction, reverse repurchase transaction, consignment, lease, swap, hedge transaction, deposit, loan, option, allocated transaction, unallocated transaction, or any other similar agreement;
- (B) any combination of agreements or transactions referred to in subparagraphs (A) and (C);
- (C) any option to enter into an agreement or transaction referred to in subparagraph (A) or (B); [or]
- (D) a master agreement that provides for an agreement or transaction referred to in subparagraph (A), (B), or (C), together with all supplements to any such master agreement, without regard to whether such master agreement provides for an agreement or transaction that is not a forward contract under this paragraph, except that such master agreement shall be considered to be a forward contract under this paragraph only with respect to each agreement or transaction under such master agreement that is referred to in subparagraph (A), (B), or (C) . . .

11 U.S.C. § 101(25). (A copy of the full text of section 101(25) showing the blacklined changes to this section under BAPCPA is attached hereto as Exhibit D.) In explaining the "forward contracts" to be protected under the Code, the legislative history to the 1990 Amendments states:

The primary purpose of a forward contract is to hedge against possible fluctuations in the price of a commodity. This purpose is

-

⁹ The Forward Contracts would qualify as "swap agreements" under the former definition of "swap agreement" as well.

financial and risk-shifting in nature, as opposed to the primary purpose of an ordinary commodity contract, which is to arrange for the purchase and sale of the commodity. If the price of a commodity – such as crude oil or soy beans – rises or falls on some future date, the buyer or seller can minimize the risk involved through the use of forward contracts to offset the fluctuation in price from the date of the agreement to the actual date of transfer or delivery.

H.R. REP. No. 101-484, at 4 (1990), reprinted in 1990 U.S.C.C.A.N. 223, 226.

In the instant case, the Forward Contracts meet the Code definition of "forward contract." First, in applying this definition of "forward contract," courts have deemed natural gas to be a commodity under this definition. See, e.g., BCP Liquidating LLC v. Bridgeline Gas Mktg., LLC (In re Borden Chemicals & Plastics Operating L.P.), 336 B.R. at 218 ("This Court believes that, at this point in time, it can hardly be questioned that natural gas is a commodity under the Code."); In re Mirant, 310 B.R. 548, 565 (Bankr. N.D. Tex. 2004)("natural gas is a commodity").

Additionally, since the Forward Contracts provided for delivery of natural gas at a fixed price during a future Delivery Period (as defined in the Base Contract), the Forward Contracts have a maturity date beyond two days after the date of the contract. <u>BCP Liquidating LLC v. Bridgeline Gas Mktg., LLC (In re Borden Chemicals & Plastics Operating L.P.)</u>, 336 B.R. at 223 (contract met definition of forward contract where natural gas delivered more than two days after contract date).

Specifically, section 1.2 of the Base Contract provides for gas and purchase transactions for a certain "Delivery Period" to be effectuated by an EDI (electronic data interchange, as specifically defined in the Base Contract) transmission or telephone conversation, which may then be confirmed by a written confirmation. This oral or EDI transmission will provide for the purchase price and "Delivery Period" for the natural gas

by the Debtor to Defendant. The Debtor then delivers the natural gas in the designated "Delivery Period" at the agreed upon price and invoices Defendant for such natural gas in the month after delivery.

For example, the email dated August 12, 2005, attached to the Proof of Claim, effectuated and confirmed the sale of natural gas from the Debtor to Defendant for the period of November 2005 through March 2006 at a capped price of \$6.60 per dekatherm. The emails dated April 2, 2004 and February 25, 2005, attached hereto as Exhibits B-2 and B-3, effectuated and confirmed the sale of natural gas, respectively, for the period of April, 2004 through March, 2005 at a capped price of \$5.70 per dekatherm, plus basis, and for the period of April, 2005 through October, 2005, at a fixed price of \$5.75 per dekatherm, plus basis. In this manner, the Debtor would effectuate a transaction with Defendant by email setting forth the natural gas price and the future "Delivery Period." The Delivery Period set forth in these emails would be for multiple future months. Accordingly, the Forward Contracts between the Defendant and the Debtor provided for a maturity date well beyond the minimum two-day maturity period required under the Code definition of "forward contract."

Finally, the Base Contract itself, which is a standard form contract produced by the NAESB, provides that, "the parties agree that the transactions hereunder constitute a 'forward contract' within the meaning of the United States Bankruptcy Code and that Buyer and Seller are each 'forward contract merchants' within the meaning of the United States Bankruptcy Code." See Base Contract, ¶ 10.5. Accordingly, the parties agreed at the time of entering into the Base Contract that the Forward Contracts were "forward contracts."

Therefore, by their plain terms, the Forward Contracts between Defendant and the Debtor meet the Code definition of "forward contracts." However, even if there was any question of whether the Forward Contracts meet the definition of "forward contract" under section 101(25) of the Code, the Forward Contracts clearly involve forward transactions, i.e., the sale and delivery of natural gas at a fixed price during future delivery periods, such that the Forward Contracts constitute "forward agreements" as that term is used in definition of "swap agreement." 11 U.S.C. 101 (53B)(A)(i)(VII); H.R. REP. No. 109-31, Pt. 1, at 129 (2005), *reprinted in* Collier on Bankruptcy, App. Pt. 10-405 (2006) ("The use of the term 'forward' in the definition of 'swap agreement' is not intended to refer only to transactions that fall within the definition of 'forward contract.' Instead, a 'forward' transaction could be a 'swap agreement' even if not a 'forward contract.'").

Accordingly, as a matter of law, the Forward Contracts between the Debtor and Defendant are commodity forward transactions and forward agreements and therefore meet the definition of "swap agreements" under section 101(53B)(A)(i)(VII) of the Code.

ii. Defendant Is A "Swap Participant" As That Term Is Used In Section 546(g) Of The Code.

One of the requirements under section 546(g) of the Code is that the transfers be made by or to a "swap participant." The Code defines "swap participant" as "an entity that, at any time before the filing of the petition, has an outstanding swap agreement with the debtor." 11 U.S.C. § 101(53C).

Since the Forward Contracts constitute prepetition "swap agreements," it is indisputable that Defendant is a "swap participant" for purposes of the Code.

iii. Because The Transfers Were Prepetition Transfers
Made To A Swap Participant Under Or In Connection
With A Swap Agreement, Section 546(g) Prevents The
Avoidance Of The Transfers As Constructive
Fraudulent Transfers.

As noted above, Section 546(g) states, in relevant part, that, "notwithstanding section 548(a)(1)(B)," the trustee cannot avoid a transfer made to a swap participant under or in connection with any swap agreement, except under section 548(a)(1)(A). 11 U.S.C. § 546(g).

Because the Forward Contracts are "swap agreements," Defendant is a "swap participant" and the Transfers the Trustee seeks to avoid were made under and in connection with such "swap agreements," section 546(g) prevents the avoidance of the Transfers as constructive fraudulent transfers under section 548(a)(1)(B) of the Code. Therefore, the Court should dismiss in their entirety the Trustee's section 548(a)(1)(B) claims in the Complaint.

B. The Transfers Are Not Subject To Avoidance And Recovery From

Defendant As Actual Fraudulent Transfers Because Defendant Received

Such Transfers For Value And In Good Faith, Such That Defendant May

Retain Such Transfers Under Section 548(c) Of The Code.

In the Third Claim of his Complaint, the Trustee alleges that, "[p]ursuant to 11 U.S.C. §§ 548(c) and 550(a), Plaintiff may recover from the Defendant the full value of the Transfers as set forth herein for the benefit of the Debtor's estate, except only to the extent the Defendant gave value to the Debtor in exchange for such Transfers [sic]." See Complaint, ¶ 26. In this paragraph, the Trustee appears to concede (as he must)

Defendant's good faith in entering into the Forward Contracts and further concedes that, under section 548(c) of the Code, Defendant can retain the Transfers (and the value of the Transfers) to the extent of the "value" that Defendant provided to the Debtor in exchange

for the Transfers.

Section 548(c) of the Code provides Defendant with a complete defense to the Trustee's claims seeking avoidance of the Transfers as actual fraudulent transfers under section 548(a)(1)(A) of the Code. ¹⁰ Section 548(c) provides:

Except to the extent that a transfer or obligation voidable under this section is voidable under section 544, 545, or 547 of this title, a transferee or obligee of such a transfer or obligation that takes for value and in good faith has a lien on or may retain any interest transferred or may enforce any obligation incurred, as the case may be, to the extent that such a transferee or obligee gave value to the debtor in exchange for such a transfer or obligation.

11 U.S.C. § 548(c).

In another example of the Code's protection from avoidance of transactions in connection with a "swap agreement," section 548(d)(2)(D) provides a conclusive and irrebutable presumption of "value" in exchange for transfers in connection with a "swap agreement." Section 548(d)(2)(D) provides:

(2) In this section –

(D) a swap participant or financial participant that receives a transfer in connection with a swap agreement takes for value to the extent of such transfer;

11 U.S.C.§ 548(d)(2)(D). (The full text of section 548 of the Code showing the blacklined changes made to this section under BAPCPA is attached hereto as Exhibit E.) Section 548(d)(2)(D) confirms that value is given with respect to any transfers made by a debtor to a swap participant in connection with a "swap agreement." See Gredd v. Bear

-

¹⁰ To the extent section 548(c) of the Code is an affirmative defense, the Court can review this affirmative defense on this Motion since this affirmative defense appears on the face of the Trustee's Complaint (see Complaint, ¶26). Campbell v. Cathcart (In re Derivium Capital, LLC), 2006 Bankr. LEXIS 3592 (Bankr. D.S.C. Dec. 22, 2006) ("Notwithstanding the general rule that affirmative defenses should not be considered on a motion to dismiss, the Fourth Circuit allows defenses to be considered if they clearly appear on the face of the complaint.").

Stearns Securities Corp. (In re Manhattan Investment Fund Ltd.), 2007 Bankr. LEXIS 49, *35 (Bankr. S.D.N.Y. 2007) (citing to analogous provision, section 548(d)(2)(B))("There is no dispute that Bear Stearns took the transfers in question for value; however the relevant issue herein is whether or not Bear Stearns took in good faith."); COLLIER ON BANKRUPTCY, ¶ 548.09[4] (15th ed. 2006)(in discussing similar language pertaining to master netting agreements, stating: "As a consequence, much like the effect of the other clauses of section 548(d)(2), section 548(d)(2)(C) [sic] essentially provides for a statutory presumption that value is given with respect to master netting agreements with respect to overall value across each of the connected financial contracts.").

Congress' intent to create a statutory presumption of "value" given in exchange for transfers in connection with a "swap agreement" is further demonstrated by the adoption of similar language dealing with master netting agreements, added as section 548(d)(2)(E) under BAPCPA. The legislative history for this new section 548(d)(2)(E) under BAPCPA describes that the purpose of the addition is:

to provide that transfers made under or in connection with a master netting agreement may not be avoided by a trustee except where such transfer is made with actual intent to hinder, delay or defraud and not taken in good faith. This amendment provides the same protections for a transfer made under, or in connection with, a master netting agreement as currently is provided for margin payments, settlement payments and other transfers received by commodity brokers, forward contract merchants, stockbrokers, financial institutions, securities clearing agencies, repo participants, and swap participants under sections 546 and 548(d), except to the extent the trustee could otherwise avoid such a transfer made under an individual contract covered by such master netting agreement.

H.R. Rep. No. 109-31, 109th Cong., 1st Sess. 132, footnote 11 (2005), reprinted in Collier on Bankruptcy, ¶ 548.09[4] (15th ed. 2006).

Since the Transfers in the instant case were made to a "swap participant" in connection with "swap agreements," Defendant provided "value" in exchange for such Transfers and is entitled to retain the full value of such Transfers under section 548(c) of the Code, to the extent Defendant received such Transfers in good faith.

As noted above, the Trustee references section 548(c) in his Complaint. In making this reference, the Trustee does not allege that Defendant received the Transfers in bad faith, nor is there any allegation in the Complaint that Defendant acted in bad faith. Accordingly, the failure of the Trustee to allege any lack of good faith on the part of Defendant in the Complaint, despite the reference to section 548(c), demonstrates there is no material issue of fact that Defendant received the Transfers in good faith. 11

Since there is no material issue of fact as to whether Defendant received the Transfers in good faith and gave "value" in exchange for the Transfers, section 548(c) of the Code prevents the Trustee from avoiding the Transfers as actual fraudulent transfers, and the Court should dismiss the Trustee's claims under section 548(a)(1)(A).

IV. CONCLUSION

For the reasons stated above, the Court should grant the Motion, under Bankruptcy Rule 7012(b)(6), and dismiss the Trustee's claims in the Complaint seeking to avoid and recover the Transfers as fraudulent transfers, under sections 548(a)(1)(A)

.

To the extent the Court determines it must receive evidence on the Defendant's "good faith" to determine completely the Defendant's section 548(c) defense to actual fraudulent transfers, the Court should grant partial summary judgment under Rule 56(d) of the Federal Rules of Civil Procedure, made applicable by Bankruptcy Rule 7056(d), on the section 548(c) issue of "value." See Fetla's Trading Post, Inc. v. Granet (In re Fetla's Trading Post, Inc.), 2006 Bankr. LEXIS 696 (Bankr. N.D. Ill. May 4, 2006) ("Fed. R. Civ. P. 56(d) provides for the situation when judgment is not rendered upon the whole matter, but only a portion thereof. The relief sought pursuant to Rule 56(d) is styled partial summary judgment. Partial summary judgment is available to dispose of one or more counts of a complaint in their entirety. Rule 56(d) provides a method whereby a court can narrow issues and facts for trial after denying in whole or in part a motion properly brought under Rule 56."). The Defendant, of course, reserves all of its rights to dispute that the Debtor made the Transfers with actual fraudulent intent, as required by section 548(a)(1)(A).

and 548(a)(1)(B) of the Code. Alternatively, the Court should grant summary judgment in favor of Defendant as to the Trustee's claims in the Complaint seeking to avoid and recover the Transfers as fraudulent transfers, under both sections 548(a)(1)(A) and 548(a)(1)(B) of the Code. Finally, Defendant requests the Court to award it the costs and expenses, including attorneys' fees, it has incurred in defending against this adversary proceeding, and to grant Defendant such other and further relief the Court deems appropriate.

Respectfully submitted, this the 15th day of February, 2007.

McGUIREWOODS LLP

By: /s/ Robert A. Cox, Jr.

Thomas E. Cabaniss (NC Bar No. 25881) Dion W. Hayes (not admitted in NC) Robert A. Cox, Jr. (NC Bar No. 21998) Bank of America Corporate Center 100 North Tryon Street, Suite 2900 Charlotte, North Carolina 28202 Telephone: (704) 373-4637

Counsel to The Smithfield Packing Company, Incorporated

CERTIFICATE OF SERVICE

I hereby certify that a true copy of the foregoing MEMORANDUM IN SUPPORT OF DEFENDANT'S MOTION: (A) TO DISMISS TRUSTEE'S COMPLAINT FOR FAILURE TO STATE A CLAIM UNDER FEDERAL RULE OF BANKRUPTCY PROCEDURE 7012, OR, IN THE ALTERNATIVE, (B) FOR SUMMARY JUDGMENT UNDER FEDERAL RULE OF BANKRUPTCY PROCEDURE 7056 was sent to those parties receiving notice in the above-referenced proceeding through the Court's CM/ECF system and sent to the following parties by regular U.S. mail, this 15th day of February, 2007:

John A. Northen, Esq.
David M. Rooks, Esq.
Stephanie Osborne-Rodgers, Esq.
NORTHEN BLUE LLP
Post Office Box 2208
Chapel Hill, North Carolina 27515-2208

/s/ Robert A. Cox, Jr. Robert A. Cox, Jr.

\4424263

EXHIBIT A

FORL1B10 (Official Form 10)(4/01) UNITED STATES BANKRUPTCY	COURT	
EASTERN DISTRICT OF NORTH CAROLIN		
Name of Debtor	Case Number	the art to the same that the manager of the section
NATIONAL GAS DISTRIBUTORS, LLC fdba Paul Lawing Jr., LLC	06-00166	
See Plans and desired the see of		
restre descope domicorestence e independanti de la consecue de la	etindik isaber sananci ing kisanci.	FILED
		MAY 2 2 2006
Name of Creditor (The person or other entity to whom the debtor	☐ Check box if you are aware that	
owes money or property):	anyone else has filed a proof of	PEGGY B. DEANS, CLERK U.S. BANKRUPTCY COURT EASTERN DISTRICT OF N.C.
THE SMITHFIELD PACKING COMPANY, INCORPORATED	claim relating to your claim. Attach copy of statement giving	CASTERN DISTRICT OF N.C.
INCORPORATED	particulars.	
Name and Address where notices should be sent:	Check box if you have never received any notices from the	
c/o Robert A. Cox, Jr. MCGUIREWOODS LLP	bankruptcy court in this case. Check box if the address differs	
100 N. Tryon Street, Suite 2900 Charlotte, NC 28202-4011	from the address on the envelope	THIS SPACE IS FOR COURT USE ONLY
Charlotte, NC 28202-4011	sent to you by the court.	THIS BY ACE IS POR COOK! OSE ONL!
Telephone Number:		
Account or other number by which creditor identifies debtor:	Check here if replaces	
1. Basis for Claim	this claim amends a pre-	viously filed claim, dated
■ Goods sold	☐ Wages, salaries, and compensation (• ',
Services performed	Your SS #:	,
☐ Money loaned ☐ Personal injury/wrongful death	Unpaid compensation for services portion to	erformed
☐ Taxes ☐ Other		ate)
2. Date debt was incurred: December, 2005	3. If court judgment, date obtained:	
4. Total Amount of Claim at Time Case Filed:	\$1.098.172.85 ** See attached pages	
If all or part of your claim is secured or entitled to priority, also complete It Check this box if claim includes interest or other charges in addition to	em 5 or 6 below.	itemized statement of all
interest or additional charges.		nemized statement of an
5. Secured Claim. Check this box if your claim is secured by collateral	6. Unsecured Priority Claim. Check this box if you have an unsec	ured priority alaim
(including a right of setoff).	Amount entitled to priority \$	ured priority claim
Brief Description of Collateral: ☐ Real Estate ☐ Motor Vehicle	Specify the priority of the claim: Wages, salaries, or commissions (up	to \$4.650) * gorned within 00 days
Other	before filing of the bankruptcy petiti	on or cessation of the debtor's
Value of Collateral: \$	business, whichever is earlier - 11 U Contributions to an employee benefit	I.S.C. §507(a)(3).
	☐ Up to \$ 2,100* of deposits toward po	irchase, lease, or rental of property or
	services for personal, family, or hous Alimony, maintenance, or support or	sehold use - 11 U.S.C. §507(a)(6).
Amount of arrearage and other charges at time case filed	child - 11 U.S.C. §507(a)(7).	•
included in secured claim, if any: \$	☐ Taxes or penalties owed to governmed ☐ Other – Specify applicable paragraph	ental units - 11 U.S.C. §507(a)(8).
	*Amounts are subject to adjustment on 4 with respect to cases commenced on	/1/04 and every 3 years thereafter or after the date of adjustment.
Credits: The amount of all payments on this claim has been credited making this proof of claim.		HIS SPACE IS FOR COURT USE ONLY
8. Supporting Documents: Attach copies of supporting documents, suc	h as promissory notes, purchase	
orders, invoices, itemized statements of running accounts, contracts, couragreements, and evidence of perfection of lien. DO NOT SEND ORI	rt judgments, mortgages, security	
documents are not available, explain. If the documents are voluminous,	attach a summary	
 Date-Stamped Copy: To receive an acknowledgement of the filing of self-addressed envelope and copy of this proof of claim. 	f your claim, enclose a stamped,	
Date Sign and print the name and title, if any, of the creditor or other		
. I sign and plint the light and title. If any, of the creditor or other	r narron outhoused to £1_4L2_	
Silo / praim Ottach copy of power of attorney, if any):		
Penalty for presenting fraudulent claim: Fine of up to \$500,000 or imprison	The Smithfield Packing Co. Inc.	

ATTACHMENT TO PROOF OF CLAIM

The Smithfield Packing Company, Incorporated. ("Smithfield") purchased natural gas from National Gas Distributors, LLC, the debtor herein ("NGD"), pursuant to a series of forward contracts, as that term is defined under section 101(25) of the Bankruptcy Code and used in section 53(B) of the Bankruptcy Code. The series of forward contracts between Smithfield and NGD were governed by the terms and conditions set forth in a certain "Base Contract for Sale and Purchase of Natural Gas," a copy of which is attached hereto. The last forward contract entered into between Smithfield and NGD provided for the sale of natural gas for the period of November 2005 through March 2006 at a capped price of \$6.60, as confirmed in that email dated August 12, 2005 from Paul Lawing to Bob Miller, a copy of which is attached hereto (the "Contract").

In December, 2005, NGD notified Smithfield that NGD could no longer sell Smithfield natural gas. As a result of NGD's failure to perform under the Contract, Smithfield was required to cover the loss of natural gas and purchase replacement natural gas from other providers. Based upon NGD's breach of the Contract in its failure to deliver natural gas, Smithfield is owed damages by NGD in the amount of \$1,098,172.85, which represents the damages to Smithfield in covering the loss and obtaining replacement natural gas for the period of December, 2005 through March, 2006.

Smithfield expressly reserves the right to amend or modify this Proof of Claim at any time for any reason, including without limitation, to add additional amounts due under the Contract for any interest, costs, and expenses to the extent permitted by applicable law. Smithfield also expressly reserves its rights against any and all third parties and/or individual guarantors with respect to the obligations set forth in this Proof of Claim. Smithfield further reserves its right to assert any other claims arising post-petition against NGD.

Base Contract for Sale and Purchase of Natural Gas

	APA	dbutors, LLC			_ and Smith	iek fiet	le ino - "	nis Base Contre	>
Empodes alle	SUDT.	ributors, LLC 1314D Rasford 28305	Road		Highw	ey 87 No	effi	B) [100]	
CHARGALIS.	NC.	28305		्रा	Tar Ha	BI. NC 2	8902		
	9r:				Diana i	dumbar	0309857	720	
Contract Nur	mber,				Contro	ct Numb	<u> </u>	30	
U.S. Federal	Tax	O Number: 582	2238079		U.S. F	w numb Idaral Ta	er.		0383930
Notices:					_	SADIGITIE	ıx IV NUN	ncer:	038395p
National Gas Attn: Paul Le	<u>Distr</u>	butors, LLC			Smithfl	ald Food	s. Inc		•
Phone: 910-	484.0		x: 910-484:	6465	. Attn: <u>B</u>	ob Miller	•		
Confirmation		, Pa	A. 810-404-	8108	. Phone:	757-35	7-8128	Fax: <u>75</u>	7-357-8136
Attn:			·	·	•			San San San San	
Phone:			Fax:		Attr:			1.	- :
involces: end			· · · · · · · · · · · · · · · · · · ·		Phone:	<u> </u>		Fiex:	
National Gasti	Ölkiril	differe J.E.C.	•	•			:	en	11, 2,44
Altn: Militar	vifia		·		Similação	lo Fried	rinc.		3
O. Box Situ	51				Attn: Bo	b Miler			
hone: 930	484.0	164	Fay: 040	49.4 0464	F@ Box	9002 S	nithfield '	A 23431	
· · · · · · · · · · · · · · · · · · ·		Carrier Co.	LEK 310-	-01-N10A	Phone:	757-957	8128	Fex: 757	-357-8136
vire Transfe	Dr A	GH Numbers (() k.k. Trust - Fave	applicable	d:		5.			***
ANK: Brand	:Ban	k & Trust - Fave	tieville:		PANK	•			
BA: 053 (0)	121				BANK:				
CCT: 51948	9792				ADÓT:			 	4-7-1
THE PERMIT					AUU):	4 = 10 = :	. i		
					omer De	Telis:			ekraj Ges publishe
Transaction. Procedure	_ i	Ofall (default) William			eral Termis and Scriptoral gree to the following reult provision she Section 7.2 Payment Date	∵.⊡ del	10"" (Jai IVO(V (ded	y of Wighth folio	white Withouth of
\$48 M. 1			10 m 11 may 20		5	· . · · · · · · · · · · · · · · · · · ·	_Net 10	Us. Day of M	onlii following
ection 2.5		2 Business Day	s efter recei	tit (dafauli)	Section 7.2	TVQD	noi oi de	hefer (defeat)	
Onfirm	O	. Bucinese	Dave effer	La facione A					
Jeadine .		Business	- mid à missi	receipt:	Methed of	. · 🖟	Automa	ф.Сентпоро	use Gredit (ACH)
action 2.A.	<u> </u>				Payment		Auloma Check	ido Cleйлидно	use Gredit (ACH)
ection 2.6	0;	Sailer (default)		receipt	Payment Section 7.7		Check Netting	ied Clejiringho	US# Gredit (ACH)
ection 2.6 confining arty	□; □. ■;	Sailer (default)			Payment		Check Netting	ied Clearinghod Distilles (Versuit) Toda not apply	
ection 2.8 confirming arty	□ : □ : ■ :	Seller (default) Buyer			Payment Section 7.7 Netting		Check Netting Netting	iad Glejaringhor Politica (dataut) Ioda not popily	
ection 2.6 confirming arty ection 3.2 erformance	□ : □ : ■ :	Sailer (default)			Payment Section 7.7 Netting Section 10.3. Early Terminate		Check Netting Netting	idd Glejiringhod Ysglidd Yddiaun) Iodd rigi apply	
ection 2:8 confirming arty action 3:2 erformance bligation ote: The fall		Seller (defaun) Buyer Gover Standard Spot Price Stan	f (default) dard		Payment Section 7.7 Neiting Section 10.3. Early Farminatic Damages Section 10.3.		Automa Check Netting Netting) Early Te Early Te	ist Cleininghou Politics (Genut) Tota not apply Intration Dame mination Dame	прин Арріу (defaul ndes Do Not Apply
section 2:8 confirming serty section 3:2 erformance bligation lots: The foll of the Immedi	owing	Saller (defaun) Buyer Gover Standard Spot Price Stan Spot Price Pu preceding.	i (detault) dard blication ap		Payment Section 7.7 Netting Section 10.3. Early.Terminatic Damages Section 10.3. Other Agreemen Setoffs		Check Netting Netting Netting Early Te Early Te	idd Clejiringhou Appliles (Meaus) 1068 noi apply Trainallon Dame Taleniari Scrott	
section 2.8 continuing sarty section 3:2 eriomenca bligation orte: The foll f the limmed section 2.26 cot Pice ablication	owing	Seller (defaun) Buyer Gover Standard Spox Price Stan Spot Price Pulpreceding. Gas Daily Midpo	f (default) idard blication ag bint (default)	oplies to both	Payment Section 7.7 Neiting Section 10.3. Early Terminate Damegies Section 10.3. Other Agreement		Automa Check Netting I Netting I Early Te Early Te Other Ag	idd Clejiringhou Appliles (Meaus) 1068 noi apply Trainallon Dame Taleniari Scrott	TOTAL (default
loté: The fall f the immedi	owing	Seller (default) Buyer Gover Standard Spot Price Stan Spot Price Pulpreceding. Gas Daily Midpo	i (default) dard blication ap pint (default) nd After Dei	oplies to both.	Payment Section 7.7 Neiting Section 10.3. Early faminate Damages Section 10.3. Other Agreemen Sectors Section 14.5 Cholos Of Law Section 14.10		Automa Check Netting Netting N	idd Clejiringhou Appliles (defaut) 1068 nol apply Indication Dami mination Dami reament Selofi reament Selofi Carolina	rgeriApply (default rges Do Not Apply e Apply (default) a Do Not Apply
Section 2.8 Confirming Party Section 3.2 Section 3.2 Section 3.2 Section 3.2 Section 3.2 Section 2.26 Section 2.26 Section 3.26 Section 8 Execution 8 Execution 8 Execution 8 Execution 8	owing	Saller (defaun) Buyer Gover Standard Spot Price Stan Spot Price Pu preceding. Gas Daily Midpo Buyer Pays At au uit) Saller Pays Beto	i (default) dard blication ap bint (default) nd After Dei	oplies to both.	Payment Section 7.7 Netting Section 10.3. Early, Termination Damages Section 10.3. Other Agreement Section 14.5 Choice Of Law		Automa Check Netting Netting N	idd Clejiringhou hpfilles (disaut) fode not apply frakation Demi mination Demi resment Seloff resment Seloff	ICES APPIV (default Ries Do Not Apply & Apply (default) & Do Not Apply
Section 2.8 Confirming Party Section 3.2 Section 3.2 Section 3.2 Section 3.2 Section 3.2 Section 2.26 Section 2.26 Section 3.26 Section 8 Execution 8 Execution 8 Execution 8 Execution 8	owing	Seller (default) Buyer Gover Standard Spot Price Stan Spot Price Pulpreceding. Gas Daily Midpo	i (default) dard blication ap bint (default) nd After Dei	oplies to both.	Payment Section 7.7 Neiting Section 10.3. Early faminate Damages Section 10.3. Other Agreemen Sectors Section 14.5 Cholos Of Law Section 14.10		Automa Check Netting Netting N	idd Clejiringhou Appliles (defaut) 1068 nol apply Indication Dami mination Dami reament Selofi reament Selofi Carolina	rgeriApply (default rges Do Not Apply e Apply (default) a Do Not Apply
Section 2.8 Confirming Party Section 3:2 Section 3:2 Section 3:2 Section 3:2 Section 2.26 Section 2.26 Section 8 Exes Especial Provis Addendum(s):	owing	Seller (defaun) Buyer Gover Standard Spot Price Stan Spot Price Pulpreceding. Gas Daily Midpo Buyer Pays At au uit) Seller Pays Befo	f (default) Idead Iblication ag Dint (default) Ind After Del Ind After Del Its and At Del Its ched:	op/les to both ivery Point elivery Point	Payment Section 7.7 Netting Section 10.3. Early:Terminatic Damages Section 10.3. Other Agreemen Setolis Section 14.5 Cholos Of Law Section 14.10 Confidentiality		Automa Check Netting Netting N	idd Clejiringhou Appliles (defaut) 1068 nol apply Indication Dami mination Dami reament Selofi reament Selofi Carolina	rgeriApply (default rges Do Not Apply e Apply (default) a Do Not Apply
Section 2.8 Confirming Party: Section 3:2 Section 3:2 Section 3:2 Section 3:2 Section 2.26 Section 2.26 Section 8 Exes Special Provia Addendum(s): VITNESS Wilcons Gas Did	owing detaily (defa	Spiler (defaun) Buyer Gover Standard Spot Price Stan Spot Price Pulpreceding. Gas Daily Midpo Buyer Pays At au uit) Seller Pays Befor	f (default) Idead Iblication ag Dint (default) Ind After Del Ind After Del Its and At Del Its ched:	op/les to both ivery Point elivery Point	Payment Section 7.7 Netting Section 10.3. Early:Terminatic Damages Section 10.3. Other Agreemen Setolis Section 14.5 Choice Of Law Section 14.10 Confidentiality	in C	Automa Check Netting Netting Netting Early Te Early Te Other Ag North Confiden	idd Geilininghod Profiles (desuit) Post roll apply Prilestion Demi mination Demi mination Demi resment Selofi resment Selofi resment Selofi Carolina dality applies (diality does not	ICES APPIV (default Ries Do Not Apply & Apply (default) & Do Not Apply
Section 2.8 Confirming Party: Section 3:2 Section 3:2 Section 3:2 Section 2.26 Section 2.26 Section 8 Sect	owitical control of the control of t	Splier (default) Buyer Gover Standard Spot Price Stan Spot Price Pu preceding. Gas Daily Midpo Buyer Pays At au uit) Splier Pays Befor Number of sheets OF, the parties in	f (default) Idead Iblication ag Dint (default) Ind After Del Ind After Del Its and At Del Its ched:	op/les to both ivery Point elivery Point	Payment Section 7.7 Netting Section 10.3. Early:Terminatic Damages Section 10.3. Other Agreemen Setolis Section 14.5 Choice Of Law Section 14.10 Confidentiality	and	Automa Check Netting Netting Netting Early Te Early Te Other Ag North Confiden	idd Clejiringhou Appliles (defaut) 1068 nol apply Indication Dami mination Dami reament Selofi reament Selofi Carolina	rgeriApply (default rges Do Not Apply e Apply (default) a Do Not Apply
Section 2.8 Confirming Serty: Section 3.2 Sertointained Section 2.26 Section 2.26 Section 8 Sect	owing defends	Splier (default) Buyer Gover Standard Spot Price Stan Spot Price Pu preceding. Gas Daily Midpo Buyer Pays At an uit) Saller Pays Before Number of sheets OF, the parties in	f (default) Idead Iblication ag Dint (default) Ind After Del Ind After Del Its and At Del Its ched:	op/les to both ivery Point elivery Point	Payment Section 7.7 Netting Section 10.3. Early. Terminatic Damages Section 10.3. Steric Agreement Section 14.5 Choice of Law Section 14.10 Confidentiality Isse Contract in des	in Co	Automa Check Netting : Netting : Netting : Early Te Other Ag Other Ag North : Confiden Confiden	idd Geilininghod Profiles (desuit) Post roll apply Prilestion Demi mination Demi mination Demi resment Selofi resment Selofi resment Selofi Carolina dality applies (diality does not	rgeriApply (default rges Do Not Apply e Apply (default) a Do Not Apply
Section 2.8 Confirming Party Section 3:2 Section 3:2 Section 3:2 Section 3:2 Section 3:2 Section 2.26 Section 2.26 Section 3:2	owing defends	Splier (default) Buyer Gover Standard Spot Price Stan Spot Price Pu preceding. Gas Daily Midpo Buyer Pays At an uit) Saller Pays Before Number of sheets OF, the parties in	f (default) Idead Iblication ag Dint (default) Ind After Del Ind After Del Its and At Del Its ched:	op/les to both ivery Point elivery Point	Payment Section 7.7 Neiting Section 10.3. Early Termination Darmegies Section 10.3. Other Agreement Section 14.5 Choice Of Law Section 14.10 Confidentiality use Contract in de Smithfield Party Name	aplicate.	Automa Check Netting Netting Netting Early Te Early Te Other Ag North Confiden	idd Geilininghod Profiles (desuit) Post roll apply Prilestion Demi mination Demi mination Demi resment Selofi resment Selofi resment Selofi Carolina dality applies (diality does not	rgeriApply (default rges Do Not Apply e Apply (default) a Do Not Apply
Section 2.8 Confirming Party: Section 3:2 Section 3:2 Section 3:2 Section 3:2 Section 2.26 Section 2.26 Section 8 Exes Special Provia Addendum(s): VITNESS Wilcons Gas Did	owing defends	Splier (default) Buyer Gover Standard Spot Price Stan Spot Price Pu preceding. Gas Daily Midpo Buyer Pays At an uit) Saller Pays Before Number of sheets OF, the parties in	f (default) Idead Iblication ag Dint (default) Ind After Del Ind After Del Its and At Del Its ched:	op/les to both ivery Point elivery Point	Payment Section 7.7 Netting Section 10.3. Early. Terminatic Damages Section 10.3. Steric Agreement Section 14.5 Choice of Law Section 14.10 Confidentiality Isse Contract in des	aplicate.	Automa Check Netting : Netting : Netting : Early Te Early Te Other Ag Other Ag North : Confiden Confiden	idd Geilininghod Profiles (desuit) Post roll apply Prilestion Demi mination Demi mination Demi resment Selofi resment Selofi resment Selofi Carolina dality applies (diality does not	rgeriApply (default rges Do Not Apply e Apply (default) a Do Not Apply

NAESB Standard 6.3.1 April 19, 2002

General Terms and Conditions Base Contract for Sale and Purchase of Natural Gas

SECTION 1. PURPOSE AND PROCEDURES

These General Terms and Conditions are intended to facilitate purchase and sale transactions of Gas on a Firm or Interruptible basis. "Buyer" refers to the party receiving Gas and "Seller" refers to the party delivering Gas. The entire agreement between the parties shall be the Contract as defined in Section 2.7.

The parties have selected either the "Oral Transaction Procedure" or the "Written Transaction Procedure" as indicated on the

Oral Transaction Procedure:

The parties will use the following Transaction Confirmation procedure. Any Gas purchase and sale transaction may be 1.2 Ine parties will use the following transaction Confirmation procedure. Any Gas purchase and sale transaction may be affectuated in an EDI transmission of telephone conversation with the offer and acceptance constituting the agreement of the parties. The parties shall be legally bound from the time they so agree to transaction terms and may each rely thereon. Any such transaction shall be considered a "writing" and to have been "signed". Notwithstanding the foregoing sentence, the parties agree that Confirming-facsimility, and the other party may, confirm a telephosic transaction by sending the other party is "transaction Confirmation by facsimility. EDI- or mutually agreeable electronic means; within three Business Days of a transaction congress by this Section 1.2 (Oral Transaction Development of the Investment of Investm Transaction: Procedure) provided that the failure to send a Transaction Confirmation shall not invalidate the oral agreement of the parties. Confirming Party adopts its confirming letterhead, or the like; as its signature on any Transaction Confirmation as the identification and authentication of Confirming Party. If the Transaction Confirmation cantains any provisions other than takes relating to the commercial terms of the transaction (i.e., price; quantity, performance obligation, delivery point, period of delivery and/or transportation conditions); which modify or supplement the Base Contract or General Terms and Conditions of this Contract (e.g., arbitration or additional representations and warranties), such provisions shall not be deemed to be accepted pursuant to Section 1.3 but must be expressly agreed to by both parties; provided that the foregoing shall not invalidate any transaction agreed to by the

Written Transaction: Procedure:

- 1.2. The parties will tase the following transaction:Confirmation procedure. Should the parties come to an agreement regarding a Gas purchase and sale transaction for a particular Delivery Period, the Confirming Party shall, and the other party hay record that agreement on a Transaction Confirmation and communicate such Transaction Confirmation by facilities. EDI or intitually agreeable electronic means, to the other party by the close of the Business Day following the date of agreement. The parties acknowledge that their agreement will not be binding until the exchange of monconflicting Transaction Confirmations or the passage of the Confirm Deadline without objection from the receiving party, as provided in Section 1.3.
- 1.3. If a sending party's Transaction Commission is materially different from the receiving party's understanding of the agreement referred 1.3. If a sending party's Transaction Confirmation is materially different from the receiving party's understanding of the agreement referred to in Section 1.2, such receiving party shall notify the sending party via facetimile, EDI or ritutually agreeable electronic means by the Confirm Deadline, unless such receiving party has proviously sent a Transaction Confirmation to this sending party. The failure of the receiving party to so notify this sending party in writing by the Confirm Deadline constitutes the receiving party's agreement to the receiving party to described in the sending party's Transaction Confirmation. If there are any material differences between timely sent Transaction governing the same transaction, then neither Transaction Confirmation shall be binding until or unless such differences are regalized including the use of any evidence; that clearly resolves the differences in the Transaction Confirmations. In the event of a confict among the terms of the parties which may be evidenced by a recorded conversalized. Where the parties have selected the Oral Transaction Procedure of the Base Contract. (iii the Contract. (ii conversation. Where the parties have selected the Oral Transaction Procedure of the Base Contract, (iii) the Base Contract, (iii) the Base Contract, (iii) the Base Contract, (iii) the Base Contract and (iv) these General Terms and Contracts, the terms of the documents shall govern in the priority listed in this sentence.
- The parties agree that each party may electronically record all telephone conversations with respect to this Contract between their respective employees, without any special or further notice to the other party. Each party shall obtain any necessary consent of its agents and employees to such recording. Where the parties have selected the Oral Transaction Procedure in Section 1.2 of the Base Contract, the parties agree not to centest the validity or enforceability of telephonic recordings entered into in accordance with the requirements of this Base Contract. However, nothing herein shall be construed as a waiver of any objection to the admissibility of such evidence.

SECTION 2. **DEFINITIONS**

The terms set forth below shall have the meaning ascribed to them below. Other terms are also defined elsewhere in the Contract and shall have the meanings ascribed to them herein.

- 2.1. "Alternative Damages" shall mean such damages, expressed in dollars or dollars per MMBtu, as the parties shall agree upon in the Transaction Confirmation, in the event either Seller or Buyer fails to perform a Firm obligation to deliver Gas in the case of Seller or to receive Gas
- "Base Contract" shall mean a contract executed by the parties that incorporates these General Terms and Conditions by reference; that specifies the agreed selections of provisions contained herein; and that sets forth other information required herein and any Special Provisions and addendum(s) as identified on page one.
- "Entitish thermal unit" or "Bitu" shall mean the International BTU, which is also called the Btu (IT).

- 2.4. "Business Day" shall mean any day except Saturday, Sunday or Federal Reserve Bank holidays.
- 2.5. "Confirm Deadline" shall mean 5:00 p.m. in the receiving party's time zone on the second Business Day following the Day a Transaction Confirmation is received or, if applicable, on the Business Day agreed to by the parties in the Base Contract; provided, if the Transaction Confirmation is time stamped efter 5:00 p.m. in the receiving party's time zone, it shall be deemed received at the opening of the next Business Day.
- 2.6. "Confirming Party" shall mean the party designated in the Base Contract to prepare and forward Transaction Confirmations to the other party.
- 2.7. "Contract" shall mean the legality binding relationship established by (i) the Base Contract, (ii) any and all binding Transaction Confirmations and (iii) where the parties have selected the Oral Transaction Procedure in Section 1.2 of the Base Contract, any and all transactions that the parties have entered into through an EDI transmission or by telephone, but that have not been confirmed in a binding Transaction Confirmation.
- 2.8. "Contract Price" shall mean the amount expressed in U.S. Dollars per MMBtu to be paid by Buyer to Seller for the purchase of Gas as agreed to by the parties in a transaction.
- 2.9. "Contract Quantity" shall mean the quantity of Gas to be delivered and taken as agreed to by the parties in a transaction.
- 2.10. "Cover Standard", as referred to in Section 3.2, shall mean that if there is an unexcused failure to take or deliver any quantity of Gas pursuant to this Contract, then the performing party shall use commercially reasonable efforts to (i) if Buyer is the performing party, obtain Gas, (or an alternate fuel if elected by Buyer and replacement Gas is not available); or (ii) if Sellet is the performing party, sell Gas, in either case, at a price reasonable for the delivery of production area, as applicable, completely with: the amount of notice provided by the nonperforming party; the immediacy of the Buyer's Gas consumption heeds or Sellet's Gas sales requirements, as applicable; the quantities involved; and the anticipated leggth of failure by the nonperforming party.
- 2.11. "Gredit Support:Obligation(s)" shall mean any obligation(s) to provide or establish credit support for, or on behalf of, a party to this Contract such as an irrevocable statidby letter of credit, a margin agreement, a prepayment, a security interest in an asset, a performance bond, guaranty, or other good and sufficient security of a continuing nature.
- 2.12. "Day" shall mean a period of 24 consecutive hours, coextensive with a "day" as defined by the Receiving Transporter in a particular transporter.
- 2.13. Delivery Period shall be the period during which deliveries are to be made as agreed to by the parties in a transaction.
- 2.14. "Delivery Point(s)" shall mean such point(s) as are agreed to by the parties in a transaction.
- 2.15. "EDI" shall mean an electronic data interchange pursuant to an agreement entered into by the parties, specifically relating to the communication of Transaction Confirmations under this Contract.
- 2.16. "EFF" shall mean the purchase, sale or exchange of natural Gas as the "physical" side of an exchange for physical transaction involving gas futures centracts. EFP shall incorporate the meaning and remadles of "Firm", provided that a party's excuse for nonperformance of its obligations to deliver or receive Gas will be governed by the rules of the relevant futures exchange regulated under the Commodity Exchange Act.
- 2.17. Firm shall mean that either party may interrupt its performance without liability only to the extent that stick performance is prevented for reasons of Force Majeure; provided, however, that during Force Majeure interruptions are party invoking Force Majeure may be responsible for any imbalance Charges as set forthin Section 4.3 related to its interruption after the nomination female to the Transporter and until the charge in deliveries and/or receipts is confirmed by the Transporter.
- 2.18. "Gae" shall mean any mixture of hydrocarbons and noncombustible gases in a gaseous state consisting primarily of methane.
- 2.19. "Imbalance Charges" shall mean any fees, penalties, costs or charges (in cash or in kind) assessed by a Transporter for failure to satisfy the Transporter's balance and/or nomination requirements.
- 2. 20. ***Interruptible* shall mean that either party may interrupt its performance at any time for any reason, whether or not caused by an event of Force Majeure, with no liability except such interrupting party may be responsible for any impalance Charges as set forth in Section 4.3 related to as interruption after the nomination is made to the Transporter and until the change in deliveres and/or receipts is confirmed by Transporter
- 2.21. "MMBtu" stial mean one million British thermal units, which is equivalent to one dekatherm.
- 2.22. "Month" shall mean the period beginning on the first Day of the calendar month and ending immediately prior to the commencement of the first Day of the next calendar month.
- 2.23. "Payment Date" shall mean a date, as indicated on the Base Contract, on or before which payment is due Seller for Gas received by Buyer in the previous Month.
- 2.24. "Receiving Transporter" shall mean the Transporter receiving Gas at a Delivery Point, or absent such receiving Transporter, the Transporter delivering Gas at a Delivery Point.
- 2.25. "Scheduled Gas" shall mean the quantity of Gas confirmed by Transporter(s) for movement, transportation or management.
- 2.26. "Spot Price" as referred to in Section 3.2 shall mean the price listed in the publication indicated on the Base Contract, under the listing applicable to the geographic location closest in proximity to the Delivery Point(s) for the relevant Day; provided, if there is no single price published for such location for such Day, but there is published a range of prices, then the Spot Price shall be the average

of such high and low prices. If no price or range of prices is published for such Day, then the Spot Price shall be the average of the following: (i) the price (determined as stated above) for the first Day for which a price or range of prices is published that next precedes the relevant Day; and (ii) the price (determined as stated above) for the first Day for which a price or range of prices is published that next follows the relevant Day.

- 2.27. "Transaction Confirmation" shall mean a document, similar to the form of Exhibit A, setting forth the terms of a transaction formed pursuant to Section 1 for a particular Delivery Period.
- 2.28. "Termination Option" shall mean the option of either party to terminate a transaction in the event that the other party fails to perform a Firm obligation to deliver Gas in the case of Selicit or to receive Gas in the case of Buyer for a designated number of days during a period as specified on the applicable Transaction Confirmation.
- 2.29. "Transporter(s)" shall mean all Gas gathering or pipeline compenies, or local distribution compenies, acting in the capacity of a transporting Gas for Seller or Buyer upstream or downstream, respectively, of the Delivery Point pursuant to a particular transaction.

SECTION 3. PERFORMANCE OBLIGATION

3.1. Saller agrees to sell and deliver, and Buyer agrees to receive and purchase, the Contract Quantity for a particular transaction in accordance with the terms of the Contract. Sales and purchases will be on a Firm or interruptible basis, as agreed to by the parties in a transaction.

The parties have selected either the "Cover Standard" or the "Spot Price Standard" as indicated on the Base Contract

Cover Ständard:

3.2. The sole and exclusive remedy of the parties in the event of a breach of a Firm obligation to deliver or receive Gas shall be recovery of the following: (i) in the event of a breach by Saller on any Day(s), payment by Seller to Buyer in an amount equal to the positive difference, if any, between the purchase price paid by Buyer utilizing the Cover Standard and the Contract Price at justed for commercially reasonable differences in transportation exists to or from the Delivery Point(s), multiplied by the difference the Contract Quartity and the quantity actually delivered by Seller for such Day(s); or (ii) in the event of a price and the point of the positive difference, if any, between the Contract Price and the price received by Seller utilizing the Cover Standard for the resals of such Ges, adjusted for commercially reasonable differences in transportation costs to or from the Delivery Point(s), multiplied by the difference between the Contract Quartity actually (alkentic) Buyer for such Day(s); or (iii) in the event that Buyer has used commercially reasonable efforts to replace the Gas of Seller has used commercially reasonable afforts to sell the Gas to a third party, and no such replacement or sale is available, they the sole and exclusive reinfied, of the performing party shall be any untavorable difference between the Contract Price and the Spot Price, adjusted for such transportation to the applicable Delivery Point, multiplied by the difference between the Contract Price and the Spot Price, adjusted for such transportation to the applicable Delivery Point, multiplied by the difference between the Contract Price and the Spot Price, adjusted for such transportation to the applicable Delivery Point, multiplied by the difference between the Contract Price and the Spot Price, adjusted for such transportation to the applicable Delivery Point, multiplied by the difference between the Contract Price and the Spot Price, adjusted for such a private by Seller and received by Buyer for such Day(s). Imbaliance

Spot Price: Standard:

- 3.2. This sole and exclusive remedy of the parties in the event of a breach of a Firm obligation to deliver or receive Eas shall be recovery of the following: (i) in the event of a breach by Seller on any Day(s), payment by Seller to Buyer in an amount equal to the difference between the Contract Quantity and the actual quantity delivered by Seller and received by Buyer for such Day(s), multiplied by the positive difference, if any, obtained by subtracting the Contract Price from the Spot Price; or (ii) in the event of a greach by Buyer on any Day(s), payment by Buyer to Seller in an amount equal to the difference between the Contract Quantity and the actual quantity delivered by Seller and received by Buyer for such Day(s), multiplied by the positive difference, it any, obtained by subtracting the applicable Spot Price from the Contract Price. Imbalance Charges shall not be recovered under this Section 3.2. But Seller and/or Buyer shall be responsible for Imbalance Charges, if any, as provided in Section 4.3. The amount of such unfavorable difference shall be payable five Business Days after presentation of the performing party's invoice, which shall set forth the basis upon which such amount was calculated.
- 3.3. Notwithstanding Section 3.2, the parties may agree to Alternative Damages in a Transaction Confirmation executed in writing by both parties.
- 3.4. In addition to Sections 3.2 and 3.3, the parties mey provide for a Termination Option in a Transaction Confirmation executed in writing by both parties. The Transaction Confirmation containing the Termination Option will designate the length of nonperformance triggering the Termination Option and the procedures for exercise thereof, how damages for nonperformance will be compensated, and how illquidation costs will be calculated.

SECTION 4. TRANSPORTATION, NOMINATIONS, AND IMBALANCES

- 4.1. Seller shall have the sole responsibility for transporting the Gas to the Delivery Point(s). Buyer shall have the sole responsibility for transporting the Gas from the Delivery Point(s).
- 4.2. The parties shall coordinate their nomination activities, giving sufficient time to meet the deadlines of the affected Transporter(s). Each party shall give the other party timely prior Notice, sufficient to meet the requirements of all Transporter(s) involved in the transaction, of the quantities of Gas to be delivered and purchased each Day. Should either party become aware that actual deliveries at the Delivery Point(s) are greater or lesser than the Scheduled Gas, such party shall promptly notify the other party.

4.3. The parties shall use commercially reasonable efforts to avoid imposition of any imbalance Charges. If Buyer or Seller receives as invoice from a Transporter that includes imbalance Charges, the parties shall determine the validity as well as the cause of such imbalance Charges. If the imbalance Charges were incurred as a result of Buyer's receipt of quantities of Gas greater than or less than the Scheduled Gas, then Buyer shall pay for such imbalance Charges were incurred as a result of Seller's delivery of quantities of Gas greater than or less than the Scheduled Gas, then Seller shall pay for such imbalance Charges or reimburse Buyer for such imbalance Charges paid by Buyer.

SECTION 5. QUALITY AND MEASUREMENT

All Gas delivered by Seller shall meet the pressure, quality and heat content requirements of the Receiving Transporter. The unit of quantity established procedures of the Receiving Transporter. When the established procedures of the Receiving Transporter.

SECTION 6. TAXES

The parties have selected either "Buyer Pays At and After Delivery Point" or "Saller Pays Bafors and At Delivery Point" as indicated on the Base Contract.

Buyer Pays At and After Delivery Point:

Seller shall pay or cause to be paid all taxes, fees, levies, penalties, licenses or charges imposed by any government authority ("Taxes") on or with respect to the Gas prior to the Delivery Point(s). Buyer shall pay or cause to be paid all Taxes on or with respect to the Gas at the Delivery Point(s) and all Taxes effect the Delivery Point(s). If a party is required to remit or pay Taxes that are the other party's responsibility hereunder, this party responsibility from any responsibility from any account taxes or charges shall furnish the other party any necessary documentation thereof.

Seller Pays: Before and At Delivery Point:

Seller shall pay or cause to be paid all taxes; fees, levies, panalties, licenses or charges imposed by any government authority. (Taxes') on or with respect to the Ges after the Delivery Point(s) and all Taxes at the Delivery Point(s). Buyer shall pay or cause to be paid all Taxes on or with respect to the Ges after the Delivery Point(s). If a party is required to remit or pay Taxes that are the other party's responsibility hereunder, the party responsible for such Taxes shall promptly relimburse the other party for such Taxes. Any party entitled to an exemption any such Taxes or charges shall furnish the other party any necessary documentation thereof.

SECTION 7. BILLING, PAYMENT, AND AUDIT

- 7.1. Seller shall invoke Buyer for Gas delivered and received in the preceding Month and for any other applicable charges, providing supporting documentation acceptable in indistry practice to support the amount charged. If the actual quantity delivered is not known by the billing date, billing will be prepared based on the quantity of Schedulet Gas. The invoked quantity will then be adjusted to the actual quantity on the following Month's billing of as soon thereafter as actual delivery information is available.
- 7.2. Buyer shall remit the amount due under Section 7.1 in the manner specified in the Base Contract in impreciately available funds, on or before the later of the Rayment Date or 10 Bays after receipt of the invoice by Buyer, provided that if the Payment Date is not a Business Day, made in accompance with this Section 7.2.
- 7.3. In the event payments become due pursuant to Sections 3.2 or 3.3, the performing party may submit ain lavoice to the nonperforming party for an accelerated payment setting forth the basis upon which the invoiced amount was calculated. Payment from the nonperforming party will be due five Business Days after receipt of invoice.
- 7.4. If the invoiced party, in good faith, disputes the amount of any such invoice or any partitioners, such invoiced party will pay such amount as it concedes to be correct; provided, however, if the invoiced party disputes the amount, due, it must provide supporting documentation acceptable in industry practics to support the amount paid or disputed. In the event the parties are unable to resolve such dispute, either party may pursue any remedy available at law or in equity to enforce its rights pursuant to this Section.
- 7.5. If the invoiced party fails to remit the full amount payable when due, interest on the unpaid portion shall accrue from the date of payment at a rate equal to the lower of () the then effective prime rate of interest published under "Money Rates" by The Wall Street Journal, plus two percent per annum; or (ii) the maximum applicable lawful interest rate.
- 7.6. A party shall have the right, at its own expense, upon reasonable Notice and at reasonable times, to examine and audit and to obtain copies of the relevant portion of the books, records, and telephone recordings of the other party only to the extent reasonably necessary to verify the accuracy of any statement, charge, payment, or computation made under the Contract. This right to examine, audit, and to obtain copies shall not be available with respect to proprietary information not directly relevant to transactions under this Contract. All involves and billings shall be conclusively presumed final and accurate and all associated claims for under- or overpayments shall be deemed walved unless such involves or billings are objected to in writing, with adequate explanation and/or documentation, within two years after the Month of Gas delivery. All retroactive adjustments under Section 7 shall be paid in full by the party owing payment within 30 Days of Notice and substantiation of such inaccuracy.
- 7.7. Unless the parties have elected on the Base Contract not to make this Section 7.7 applicable to this Contract, the parties shall net all undisputed amounts due and owing, and/or past due, arising under the Contract such that the party owing the greater amount shall make a single payment of the net amount to the other party in accordance with Section 7; provided that no payment required to be made pursuant to the terms of any Credit Support Obligation or pursuant to Section 7.3 shall be subject to netting under this Section. If the parties have executed a separate netting agreement, the terms and conditions therein shall prevail to the extent inconsistent

SECTION B. TITLE, WARRANTY, AND INDEMNITY

- Unless otherwise specifically agreed, little to the Gas shall pass from Seller to Buyer at the Delivery Point(s). Seller shall have responsibility for and assume any liability with respect to the Gas prior to its delivery to Buyer at the specified Delivery Point(s). Buyer shall have responsibility for and any liability with respect to said Gas after its delivery to Buyer at the Delivery Point(s).
- 8.2. Selier warrants that it will have the right to convey and will transfer good and merchanizable title to all Gas sold hereunder and delivered by it to Buyer, free and clear of stiffiens, encumbrances, and claims. EXCEPT AS PROVIDED IN THIS SECTION 8.2 AND IN SECTION 14.8, ALL OTHER WARRANTIES, EXPRESS OR IMPLIED, INCLUDING ANY WARRANTY OF MERCHANTABILITY OR OF FITNESS FOR ANY PARTICULAR PURPOSE, ARE DISCLAIMED.
- 8.3. Seller agrees to indemnify Buyér and save it harmless from all losses, liabilities or claims including reasonable attorneys' (ees and costs of court ("Claims"), from any and all persons, arising from or out of claims of title, personal injury or property damage from said Gas or other charges thereon which attach before title passes to Buyer. Buyer agrees to Indemnify Seller and save it harmless from all Claims, from any and all persons, atising from or out of claims regarding payment, personal injury or properly damage from said Gas or other charges thereon which attach altertitle passes to Buyer.
- Notwithstanding the other provisions of this Section 8, as between Seller and Buyer, Seller will be liable for all Cialms to the extent that such arise from the failure of Gas delivered by Selfer to meet the quality requirements of Section 5.

SECTION 9. **NOTICES**

- All Transaction Confirmations, invoices; payments and other communications made pursuant to the Base Contract ("Notices") shall be made to the addresses specified in writing by the respective parties from time to time.
- All Notices required hereunder may be sent by facsimile or mutually acceptable electronic means, a nationally recognized overnight courier service, first class mall or hand delivered:
- Notice shall be given when received on a Business Day by the addressee. In the absence of proof of the actual receipt date, the following presumptions will apply. Notices sent by facsimile shall be deemed to have been received upon the sanding party's receipt of its facsimile machine's confirmation of successful transmission. If the day on which such facsimile is regeived is not a Business Day or is after the p.m. on a Business Day, then such facsimile shall be deemed to have been received on the next following Business Day. Notice by overnight mail or courier shall be deemed to have been received on the next Business Day after twee sent or such earlier time as is confirmed by the receiving party. Notice via first class mail shall be considered delivered five Business Days after malling. 1605

SECTION 10. FINANCIAL RESPONSIBILITY.

- 10.1. If either party ("X") has reasonable grounds for insecurity regarding the performance of any obligation under this Contract (whether or not then due) by the other party ("Y") (including without limitation, the openings of a material change in the credity or things of Y). X may demand Adequate Assurance of Performance Adequate Assurance of Performance and for the term reasonably acceptable to X, including, but not implicate to a standard invesced to the second of the s letter of credit, a prepayment, a security interest in an asset or a performance bond or guaranty (dictuding the issue) of any such security):
- 10.2. In the event (each an "Event of Default") either party (the "Defaulting Party") or its guarantor shall: (i) make an assignment or 10.2. In the event each an "Event or Default") enter party (the "Defaulting Party") or its guarantor shall. (i) make an assignment or any general arrangement for the benefit of creditors; (ii) file a petition or otherwise commence, sufficieze, or acquiesce in the commencement of a proceeding or case ender any bankruptcy or similar law for the protection of creditors of have sufficient filed or proceeding commenced against it; (iii) otherwise become bankrupt or insolvent (however evidenced); (iv) be unable to pay its debts as they fall due; (v) have a receiver, provisional liquidator, conservator, custodian, trusted or other similar angles appointed with respect to it or substantially all of its assets; (vi) fall to perform any obligation to the other party with respect to any Gredit Support obligations relating to the Contract; (vii) fall to give Adequate Assufance of Performance under Section 10.1. within all hours but in less one Business Dav of a written request by the other party no viii) and have read any amount due the other rainty instruments in the before the Business Day of a written request by the other party; or (vill) not have paid any amount due the other party hereunder on a before the second Business Day following written Notice that such payment is due; then the other party (the "Noni Defaulting Party") shall have the right, at its sole election, to immediately withhold and/or suspend deliveries or payments upon Notice and/or to terminate and liquidate the transactions under the Contract, in the manner provided in Section 10.3, in addition to any and all other remedies available
- If an Event of Default has occurred and is continuing, the Non-Defaulting Party shall have the right, by Notice to the Defaulting Party, to designate a Day, no earlier than the Day such Notice is given and no later than 20 Days after such Notice is given, as an early termination date (the "Early Termination Date") for the liquidation and termination pursuant to Section 10.3.1 of all transactions under the Contract, each a "Terminated Transaction". On the Early Termination Date, all transactions will terminate, other than those transactions, if any, that may not be liquidated and terminated under applicable law or that are, in the reasonable opinion of the Non-Defaulting Party, commercially impracticable to liquidate and terminate ("Excluded Transactions"), which Excluded Transactions must be liquidated and terminated as soon thereafter as is reasonably practicable, and upon termination shall be a Terminated Transaction and be valued consistent with Section 10.3.1 below. With respect to each Excluded Transaction, its actual termination date shall be the Early Termination Date for purposes of Section 10.3.1.

The parties have selected either "Early Termination Damages Apply" or "Early Termination Damages Do Not Apply" as Indicated on the Base Contract.

Early Termination Damages Apply:

10.3.1. As of the Early Termination Date, the Non-Defaulting Party shall determine, in good faith and in a commercially reasonable manner, (i) the amount owed (whether or not then due) by each party with respect to all Gas delivered and received between the parties under Terminated Transactions and Excluded Transactions on and before the Early Termination Date and all other applicable charges relating to such deliveries and receipts (including without limitation any amounts owed under Section 3.2), for which approache charges relating to such deliveries and receipts (including without implication any smooths owed under decion 2.2), for which payment has not yet been made by the party that owes such payment under this Contract and (ii) the Market Value, as defined below, of each Terminated Transaction. The Non-Defaulting Party shall (x) liquidate and accelerate each Terminated Transaction at its Market Value, so that each amount equal to the difference between such Market Value and the Contract Value, as defined below, of such Terminated Transaction(s) shall be due to the Buyer under the Terminated Transaction(s) if such Market Value exceeds the Contract Value and to the Seller if the opposite is the case; and (y) where appropriate, discount each amount then due under clause (x) above to present value in a commercially reasonable manner as of the Early Termination Date (to take account of the period between the date of liquidation and the date on which such amount would have otherwise been due pursuant to the relevant Terminated Transactions).

For purposes of this Section 10.3.1, "Contract Value" means the amount of Gas remaining to be delivered or purchased under a transaction multiplied by the Contract Price, and "Market Value" means the amount of Gas remaining to be delivered or purchased under a transaction multiplied by the market price for a similar transaction at the Delivery Point determined by the Non-Defaulting Party in a commercially reasonable manner. To ascertain the Market Value, the Non-Defaulting Party may consider, among other valuations, any or all of the settlement prices of NYMEX Gas futures contracts, quotations from leading dealers in energy swap valuations, any or all or the settlement prices of NYMEX Das nutures contracts, quotations from leading dealers in energy swap contracts or physical gas fracing markets, similar sales or purchases and any other bona fide third-party offers, all adjusted for the tent graph of the ferm and differences in transportation costs. A party shall not be required to enter into a replacement transaction(s) in order to determine the Market Value. Any extension(s) of the term of a transaction to which parties are not bound as of the Early Termination Date (including but not limited to "evergreen provisions") shall not be considered in determining Contract Values and Market Values. For the evolution of doubt, any option pursuant to which one party has the right to extend the term of a fransaction shall be considered in determining Contract Values and Market Values. The rate of interest used in calculating net present value shall be determined by the Non-Defaulting Party in a commercially reasonable manner.

Early Termination Damages Do Not Apply:

10.3.1. As of the Early Termination Date, the Non-Defaulting Party shall determine, in good faith and in a commercially reasonable manner, the amount owed (whether or not then due) by each party with respect to all Gas delivered and received between the parties under Terminated Transactions and Excluded Transactions on and before the Early Termination Date and other applicable charges relating to such delivenes and receipts (including without limitation any amounts owed under Section 3.2), for which payment has not yet been made by the party that owes such payment under this Contract.

The parties have relected either "Other Agreement Setoffs Apply" or "Other Agreement Setoffs Do Not Apply" as indicated on the Base Contract

Other Apresment Setoffs Apply:

10.3.2. The Non-Defaulting Party shall not or aggregate, as appropriate, any and all amounts owing between the parties under Section 10.3.1, so that all such amounts are netted or aggregated to a single liquidated amount payable by one party to the other (the "Net Settlement Amount"). At its sole option and without order Notice to the Defaulting Party, the Notice Indianal Control (i) any Net Settlement Amount owed to the Non-Defaulting Party against any margin of other collateral held by it in comments with any Credit Susport Obligation relating to the Contract; or (ii) any Net Settlement Amount payable to the Defaulting Party against any amount(s) payable by the Defaulting Party to the Non-Defaulting Party under any other agreement or arrangement between the parties.

Other Agreement Setoffs Do Not Apply:

- 10.3.2. The Non-Defaulting Party shall not or aggregate, as appropriate, any and all amounts owing between the parties under Section 10.3.1, so that all such amounts are noted or aggregated to a single liquidated amount payable by one party to the other (the "Net Settlement Amount"). At its sole option and without prior Notice to the Defaulting Party, the Non-Defaulting Party may setoff any Net Settlement Amount owed to the Non-Defaulting Party against any margin or other collateral held by it in connection with any Credit Support Obligation relating to the Contract.
- 10.3.3. If any obligation that is to be included in any netting, aggregation or seloff pursuant to Section 10.3.2 is unascertained, the Non-Defaulting Party may in good faith estimate that obligation and net, apprepate or setoff, as applicable, in respect of the estimate, subject to the Non-Defaulting Party accounting to the Defaulting Party when the obligation is ascertained. Any amount not then due which is included in any netting, aggregation or setoff pursuant to Section 10.3.2 shall be discounted to net present value in a commercially reasonable manner determined by the Non-Defaulting Party.
- As soon as practicable after a liquidation, Notice shall be given by the Non-Defaulting Party to the Defaulting Party of the Net Settlement Amount, and whether the Net Settlement Amount is due to or due from the Non-Defaulting Party. The Notice shall include a written statement explaining in reasonable detail the calculation of such amount, provided that failure to give such Notice shall not affect the validity or enforceability of the liquidation or give rise to any claim by the Defaulting Party against the Non-Defaulting Party. The Net Settlement Amount shall be paid by the close of business on the second Business Day following such Notice, which date shall not be earlier than the Early Termination Date. Interest on any unpaid portion of the Net Settlement Amount shall accrue from the date due until the

date of payment at a rate equal to the lower of (i) the then-effective prime rate of interest published under "Money Rates" by The Wall Street Journal, plus two percent per annum; or (ii) the maximum applicable lawful interest rate.

- 10.5. The parties agree that the transactions hereunder constitute a "forward contract" within the meaning of the United States Bankruptcy Code and that Buyer and Seller are each "forward contract merchants" within the meaning of the United States Bankruptcy.
- 10.6. The Non-Defaulting Party's remedies under this Section 10 are the sole and exclusive remedies of the Non-Defaulting Party with respect to the occurrence of any Early Termination Date. Each party reserves to itself all other rights, setoffs, counterclaims and other defenses that it is or may be artitled to arising from the Contract.
- 10.7. With respect to this Section 10 if the parties have executed a separate netting agreement with close-out netting provisions, the terms and conditions therein shall prevail to the extent inconsistent herewith.

SECTION 11. FORCE MAJEURE

- 11.1. Except with regard to a party's obligation to make payment(s) due under Section 7, Section 10.4, and Imbalance Charges under Section 4, neither party shall be liable to the other for failure to perform a Firm obligation, to the extent such failure was caused by Force Majeure. The term "Force Majeure" as employed herein means any cause not reasonably within the control of the party claiming suspension, as further defined in Section 14.2.
- 11.2. Force Majeure shall include, but not be limited to, the following: (i) physical events such as acts of God, landslides, lightning, earthquakes, fires, storms or storm warnings, such as hurricanes, which result in evacuation of the affected area, floods, washouts, an entire geographic region, such as low temperatures which cause freezing or failure of wells or lines of pipe; (ii) weather related events affecting curtaliment of Firm fransportation and/or storage by Fransporters; (iv) acts of others such as strikes; (spectouts or other industrial disturbances; nots, sapotage, insurrections or wars; and (v) governmental actions such as acressity; (or compliance with any court order; law, statute, ordinance, regulation, of policy having the effect of law promulgated by a governmental action; such as accessity for compliance with any court order; law, statute, ordinance, regulation, of policy having the effect of law promulgated by a governmental action; such as exemple to event or occurrence once it has occurred in order to resume performance.
- 11.3. Neither party shall be entitled to the benefit of the provisions of Force Majeure to the extent performance is affected by any or all of the following circumstances: (f) the curtallment of interruptible or secondary Firm transportation unless primary; in path, Firm transportation is also curtalled; (ii) the party claiming except falled to remedy the condition and to resume the performance of such coverents or obligations with reasonable dispatch; or (iii) economic hardship, to include, without limitation. Seller's ability to said Gas at a higher or more advantageous price than the Contract Price, One regulatory agency disellewing, in whole or in part, the pass through of coets tiguiting from this Agreet fall (iv) the loss of Buyer's mairicit(s) or Buyer's mairicit(s) or Buyer's mairitity to use or resell Gas gurchased hereunder, except, in either case, as provided in Section 11.2; Or (v) the loss of fallure of Seller's gas supply or depiction of reserves, except, in either case, as provided in Section 11.2. The party claiming Force Majeure's half not be excused from its responsibility for imbalance Charges.
- 11.4. Notwithstanding anything to the contrary herein, the parties agree that the settlement of strikes, lockouts or other industrial disturbances shall be within the sole discretion of the party experiencing such disturbance.
- 11.5. The party whose performance is prevented by Force Majeure must provide Notice to the other party. Initial Notice may be given orally; however, written Notice with reasonably full particulars of the event or occurrence is required as soon as reasonably possible. Upon providing written Notice of Force Majeure to the other party, the affected party will be relieved effits obligation, from the onset of the Force Majeure event, to make or accept delivery of Gas; as applicable, to the extent and for the duration of Force Majeure, and neither party shall be deemed to have failed in such obligations to the other during such occurrence or event.
- 11.6. Notwithstanding Sections 11.2 and 11.3, the parties may agree to alternative Force Majeure provisions in a Transaction Confirmation executed in willing by both parties.

SECTION 12. TERM

This Contract may be terminated on 30 Day's written Notice, but shall remain in effect until the expiration of the latest Delivery Period of any transaction(s). The rights of either party pursuant to Section 7.6 and Section 10, the obligations to make payment hereunder, and the obligation of either party to indemnify the other, pursuant hereto shall survive the termination of the Base Contraction any transaction.

SECTION 13. LIMITATIONS

FOR BREACH OF ANY PROVISION FOR WHICH AN EXPRESS REMEDY OR MEASURE OF DAMAGES IS PROVIDED, SUCH EXPRESS REMEDY OR MEASURE OF DAMAGES SHALL BE THE SOLE AND EXCLUSIVE REMEDY. A PARTY'S LIABILITY HEREUNDER SHALL BE LIMITED AS SET FORTH IN SUCH PROVISION, AND ALL OTHER REMEDIES OR DAMAGES AT LAW OR IN EQUITY ARE WAVED. IF NO REMEDY OR MEASURE OF DAMAGES IS EXPRESSLY PROVIDED HEREIN OR IN A TRANSACTION, A PARTY'S LIABILITY SHALL BE LIMITED TO DIRECT ACTUAL DAMAGES ONLY. SUCH DIRECT ACTUAL DAMAGES SHALL BE THE SOLE AND EXCLUSIVE REMEDY, AND ALL OTHER REMEDIES OR DAMAGES AT LAW OR IN EQUITY ARE WAVED. UNLESS EXPRESSLY HEREIN PROVIDED, NEITHER PARTY SHALL BE LIABLE FOR CONSEQUENTIAL, INCIDENTAL, PUNITIVE, EXEMPLARY OR INDIRECT DAMAGES, LOST PROFITS OR OTHER BUSINESS INTERRUPTION DAMAGES, BY STATUTE, IN TORT OR CONTRACT, UNDER ANY INDIREMINITY PROVISION OR OTHERWISE. IT IS THE INTENT OF THE PARTIES THAT THE LIMITATIONS HEREIN IMPOSED ON REMEDIES AND THE MEASURE OF DAMAGES BE WITHOUT REGARD TO THE CAUSE OR CAUSES RELATED THERETO, INCLUDING THE NEGLIGENCE OF ANY PARTY, WHETHER SUCH NEGLIGENCE BE SOLE, JOINT OR CONCURRENT, OR ACTIVE OR PASSIVE.

Copyright © 2002 North American Energy Standards Board, Inc. All Rights Reserved TO THE EXTENT ANY DAMAGES REQUIRED TO BE PAID HEREUNDER ARE LIQUIDATED, THE PARTIES ACKNOWLEDGE THAT THE DAMAGES ARE DIFFICULT OR IMPOSSIBLE TO DETERMINE, OR OTHERWISE OBTAINING AN ADEQUATE REMEDY IS INCONVENIENT AND THE DAMAGES CALCULATED HEREUNDER CONSTITUTE A REASONABLE APPROXIMATION OF THE HARM OR LOSS.

SECTION 14. MISCELLANEOUS

- 14.1. This Contract shall be binding upon; and inure to the benefit of the successors, assigns, personal representatives, and heirs of the respective parties hereto, and the covenants, conditions, rights and obligations of this Contract shall run for the full term of this Contract. No assignment of this Contract, in whole or in part, will be made without the prior written consent of the non-assigning party (and shall not relieve the assigning party from liability herecinder); which consent will not be unreasonably withheld or delayed; provided, either party may (i) transfer, sell, piedge, encumber, or assign this Contract of the accounts, revenues, or proceeds hereof in connection with any financing or other financial arrangements, or (ii) transfer its interest to any parent or affiliate by assignment, merger or otherwise without the prior approval of the other party. Upon any such assignment, transfer and assumption, the transferor shall remain principally liable for and shall not be relieved of or discharged from any obligations hereunder.
- 14.2. If any provision in this Contract is determined to be invalid, void or unenforceable by any court having jurisdiction, such determination shall not invalidate, void, or make unenforceable sny other provision, agreement or covenant of this Contract.
- 14.3. No waiver of any breach of this Contract shall be held to be a waiver of any other or subsequent breach.
- 14.4. This Contract sets forth all understandings between the parties respecting each transaction subject hereto, and any prior contracts, understandings and representations, whether oral or written, relating to such transactions are merged into and superseded by this Contract and any effective transaction(s). This Contract may be amended only by a writing executed by both parties:
- 14.5. This interpretation and performance of this Contract shall be governed by the laws of the jurisdiction as indicated on the Base Contract, excluding, however, any conflict of laws rule which would apply the law of another jurisdiction.
- 14.6. This Contract and all provisions herein will be subject to all applicable and valid statutes, rules orders and equitations of any governmental authority having jurisdiction over the parties, their facilities, or Gas supply, this Contract or transaction or any provisions thereof.
- 14.7. There is no third party beneficiary to this Contract.
- 14.8. Each party to this Contract represents and warrants that it has full and complete authority to enter into and perform his Contract. Each person who executes this Contract on behalf of either party represents and warrants that it has full and complete authority to do so and that such party will be bound thereby.
- 14.9. The headings and subheadings contained in this Contract are used solely for convenience and do not constitute a part of this Contract between the parties and shall not be used to constitute to interpret the provisions of this Contract.
- 14.10. Unless the parties have elected on the Base Contract not to make this Section 14.10 applicable to this Contract, neither party shall disclose directly or indirectly without the prior written consent of the other party the terms of any transaction to a third party, other than the employees, lenders, royalty owners, counsel, accountants and other agents of the party, or prospective purchasers of all or substantially all of a party's assets or of any transaction to accountants and other agents of the party, or prospective purchasers of all or substantially all of a party's assets or of any transaction, provided such persons shall have agreed to keep such terms confidentially except (i) in order to comply with any applicable law, order, regulation, or exchange rule, (ii) to the extent necessary for the enforcement of this Contract, (iii) to the extent necessary for implement any transaction; or (iv) to the extent such information is delivered to such third party for the sole purpose of calculating a piciplished index. Each party shall motify the other party of any proceeding of which it is sware which may result in disclosure of the terms of any transaction (other than as partitled hereunder) and use reasonable efforts to prevent or limit the disclosure. The existence of this Contract is not subject to this confidentiality obligation. Subject to Section 13, the parties shall be entired to receive the experiment the experiment the transaction.

In the event that disclosure is required by a governmental body or applicable law, the party subject to such requirement may disclose the material terms of this Corpiact to the extent so required, but shall promptly notify the other party, prior to disclosure, and shall cooperate (consistent with the disclosing party a legal obligations) with the other party sent to obtain protective orders or similar restrictive with respect to such disclosure at the expense of the other party.

14.11 The parties may agree to dispute resolution procedures in Special Provisions attached to the Base Contract or in a Transaction Confirmation executed in writing by both parties:

DISCLAMER: The purposes of this Contract are to facilists trade, avoid instanders andings and make more definite the terms of contracts of purchase and sale of natural gas. Further, NAESB does not mendete that use of this Contract by any party. NAESB DISCLAMS AND EXCLUDES, AND ANY USER OF THIS CONTRACT ACKNOWLEDGES AND AGREES TO NAESB'S DISCLAMER OF, ANY AND ALL WARRANTIES, CONDITIONS OR REPRESENTATIONS, EXPRESS OR IMPLIED, ORAL OR WRITTEN, WITH RESPECT TO THIS CONTRACT OR ANY PART THEREOF, INCLUDING ANY AND ALL MAPLIED WARRANTIES OR CONDITIONS OF TITLE, NON-INFRINGEMENT, MERCHANTABILITY, OR TITLES OR SUITABILITY FOR ANY PARTICULAR PURPOSE (WHETHER OR NOT NAESB KNOWS, HAS REASON TO KNOW, HAS BEEN ADVISED, OR IS OTHERWISE IN FACT AWARE OF ANY SUCH PURPOSE), WHETHER ALLEGED TO ARISE BY LAW, BY REASON OF CUSTOM OR USAGE IN THE TRADE, OR BY COURSE OF DEALING. EACH USER OF THIS CONTRACT ALSO AGREES THAT UNDER NO CIRCUMSTANCES WILL NAESB BE LLABLE FOR ANY DIRECT, SPECIAL, INCIDENTAL, EXEMPLARY, PUNITIVE OR CONSEQUENTIAL DAMAGES ARISING OUT OF ANY USE OF THIS CONTRACT.

TRANSACTION CONFIRMATION FOR IMMEDIATE DELIVERY

EXHIBIT A

Letterhead/Lo)go			Date: Transaction (Confirmation #:	
This Transaction Confirmation terms of this Transaction Confispecified in the Base Contract.	is subject to the mation are bir	e Base Contrac iding unless dis	between Seller outed in writing	r and Buyer da within 2 Busine	led ess Days of receip	. 7 d unless otherwis
SELLER; National Gas Distributors P. O. Box 58051, 1314 D Ras			BUYER:			· · · · · · · · · · · · · · · · · · ·
Favetteville, NC 28305 Atn: Paul Lawing Phone; 910-484-9164			Altn:			
ax: #10/484-9109			Fax: Base Contra	ict No.		
ransporter: ransporter:Contract Number:			Transporter:	Contract Numb		
ontract Price: \$/MMBt				Y grings)		
elivery Period: Begin: erformance Obligation and (itity: (Spilect On	End:			Mary Mary Mary Bulk III
irm (Pixed Quentity): (MMBtus/day U EFP	1 P.	MA	ble Quantity); (Blus/day Minin	num Vig	linterrûptilb Lip to	le: MMBwarday
O par		subjectité S U Buyerer I	lBtus/day kiaoin ection 4.2. at el □ Seller	ection of		
elivery Rolint(d): a pooling point is itsed, list a	specific geogra	phic and pipelin	e location):			
pecial Goriditions:			:			
			*		**	
elleri National Gas Distribut	ors, LLC		Buyer:			1804%
lle: <u>President</u>			Title:			
ale:		······································	Date:	****		

Miller, Bob

From:

HPCBBALL@aol.com

Sent: To: Friday, August 12, 2005 2:07 PM bobmiller@smithfieldfoods.com

Subject: Winter

Bob,

To confirm our recent telephone conversation - this is to confirm a \$6.60 cap for all volumes for all the NC plants.

Thanks!

Paul Lawing, Jr. 910-484-9164 Office 910-484-9109 Fax 910-988-2320 Call Through march : 06

EXHIBIT B

UNITED STATES BANKRUPTCY COURT EASTERN DISTRICT OF NORTH CAROLINA WILSON DIVISION

NATIONAL GAS DISTRIBUTORS, LLC, Debtor. NICHARD M. HUTSON, II, TRUSTEE FOR NATIONAL AS DISTRIBUTORS, LLC, f/k/a Paul Lawing, Jr., LLC, Plaintiff, V. THE SMITHFIELD PACKING COMPANY, INCORPORATED, Defendant.	In re:)
Debtor. RICHARD M. HUTSON, II, TRUSTEE FOR) NATIONAL AS DISTRIBUTORS, LLC, No. 06-00267-8-ATS f/k/a Paul Lawing, Jr., LLC, Plaintiff,) V.	NATIONAL GAS DISTRIBUTORS, LLC,) Case No. 06-00166-8-ATS Chapter 11
NATIONAL AS DISTRIBUTORS, LLC, f/k/a Paul Lawing, Jr., LLC, Plaintiff, v. THE SMITHFIELD PACKING COMPANY, INCORPORATED, No. 06-00267-8-ATS	Debtor.))
v.) THE SMITHFIELD PACKING COMPANY,) INCORPORATED,)	NATIONAL AS DISTRIBUTORS, LLC,	•
THE SMITHFIELD PACKING COMPANY, INCORPORATED,)	Plaintiff,	,)
INCORPORATED,)	v.	,)
Defendant.)	•)))
	Defendant.	,))

AUTHENTICATION DECLARATION OF ROBERT E. MILLER

Robert E. Miller, pursuant to 28 U.S.C. § 1746, in support of the Motion (the "Motion") of The Smithfield Packing Company, Incorporated: (A) To Dismiss The Trustee's Complaint For Failure To State A Claim Under Federal Rule Of Bankruptcy Procedure 7012, or, in The Alternative, (B) For Summary Judgment Under Federal Rule Of Bankruptcy Procedure 7056, hereby declares:

1. I am Energy Manager for Smithfield Foods, Inc. In that capacity, I have the corporate authority to contract on behalf of The Smithfield Packing Company, Incorporated and Stadler's Country Hams, Inc., n/k/a The Smithfield Packing Company, Incorporated (together, "Smithfield"), wholly owned subsidiaries of Smithfield Foods, Inc., for the purchase of natural gas by these entities. I have personal knowledge of the

matters set forth in this Declaration and could competently testify thereto if called upon do to so.

- 2. I submit this Declaration in support of the Motion to authenticate a certain Base Contract and certain email exchanges by and between Smithfield and National Gas Distributors, LLC ("NGD") for the sale of natural gas by NGD to Smithfield's facilities located in Kinston, North Carolina; Wilson, North Carolina; Elon, North Carolina; and Tar Heel, North Carolina (together, the "Facilities").
- 3. Smithfield purchased natural gas from NGD for the Facilities pursuant to a series of forward contracts. The series of forward contracts between Smithfield and NGD were governed by the terms and conditions set forth in that certain Base Contract for Sale and Purchase of Natural Gas (the "Base Contract"). An authentic, true, and correct copy of the Base Contract is attached hereto as Exhibit B-1.
- 4. The sale and purchase of natural gas for the Facilities was effectuated and confirmed by emails between Smithfield and the Debtor. Attached hereto as Exhibit B-2 is an authentic, true, and correct copy of an email exchange dated April 2, 2004, which effectuated and confirmed the sale of natural gas for the period of April, 2004 through March, 2005 at a capped price of \$5.70 per dekatherm, plus basis (gas transport costs). Attached hereto as Exhibit B-3 is an authentic, true, and correct copy of an email exchange dated February 25, 2005, which effectuated and confirmed the sale of natural gas for the period of April, 2005 through October, 2005, at a capped price of \$5.75 per dekatherm, plus basis. Attached hereto as Exhibit B-4 is an authentic, true, and correct copy of an email exchange dated August 12, 2005, which effectuated and confirmed the

sale of natural gas for the period of November, 2005 through March, 2006 at a capped price of \$6.60 per dekatherm.

5. The Base Contract attached hereto as <u>Exhibit B-1</u> and the email exchanges attached hereto as <u>Exhibit B-2</u>, <u>Exhibit B-3</u>, and <u>Exhibit B-4</u> have been produced from the files of Smithfield as kept in the ordinary course of Smithfield's regularly conducted business activities.

I declare under penalty of perjury under the laws of the United States of America that the foregoing is true and correct.

Executed on February 15, 2007.

/s/ Robert E. Miller
Robert E. Miller

\4439088.1

EXHIBIT B-1

.

Base Contract for Sale and Purchase of Natural Gas

Amional Ga	M Diskilkulove I	19d into as of the	following date: _		The parties to	this Base Contr	act are the followin
P.O. BOX 5	8081 13146 B	Marini Boles		_ and <u>amiths</u>	رمدن مكانفة أيلد	- Tar Hand BM	>
	. NCC SATURE		- 41		N 87 NA-8		
Dung Numb			****		L NC 20502		
Contract Nu	mber:			Duns N	Wilher: Dicke	5738	
U.S. Federal	Tax ID Numbe	P. Mcstagnes		Contrac	il Number:		
Nation:	•			U.S. Fe	derai Tax ID N	tumber: 44.	(332930
National Gar Attn: Paul L	Distributora, L	c : : :			id Foods, inc		
hone: 910	494 9494			Alfr. Bo	th Miles		
		Fac: 910-48	1-9700	Phone:	757-957-8128	Fav. 75	7-367-8138
Onland to	唯 (,		, ==0. T2	1-301-0130
itin:				· ·			
honé:		Fac		_ Atth:			
				. Phone:		- T	
Valent and				•		đa.	
to: William	Michigan LE	<u> </u>	<u> </u>	. Smithfule	LEANGE Inc.		V 345 7
0.				Athr: Bot	Maria		, , , ,
one gra	A 10 1 10 10 10 10 10 10 10 10 10 10 10 1			. PO Sovi	Mariant Control	VALUE	
		FAX <u>910</u>	404-9105	Phone: 7	57-52-0128	Face 929	357-4/36
	A A Barrier	ers (if and heb)	iah:			1447 (0)	210(6)01/10/0
MIC HEE	Bank & Frest	- Favelleville	<u></u>	EANK:			
CT: 5				ATEA:			
er Detelle:	97977 · · ·			AQQT:			
			* * * *				
Blue Cort	خرمشها والم	Marmatan de		orgi, set ard Tentiji artičen Gree to djestičnaje Mad provinci djed Spetika 7,2			
iction 12 Indiablen Ioddur	C) William			Section 7.2 Payment Date		Water Bay of Me	
celon 2.6	40.00	3 50			1101	Bay of Mc	mili idikawing
		s Days alter rece	ript (delault)	Section 7.2	- Control of O	raheldi (dollari) aliid Glejiringhol	
Idine .	وم بنيد ت	einess Days sile	racelpt-	Method of	E Autom		
ction 2.8	D : Seller (de	4		Payment	Check	The state of the s	ST STREET, (ACH)
dinira -		caust) :		Section 7.7	Natio	The Real Property lies	
b	. 100	. 7	1.1.	Methido	E Netting	Applies (televa) Whee rightspoly	250 77
Selection State	# # # 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1					Section 1	
didn 1 4	GOVE SIZ	inderd (duteut)		Section 10.3.1		to the state of th	
Degloti	☐ Spoi Price	Standard		Bigty Termination	nyan Bankar Santar		DAY (Colour)
	والمتعارض	• •		Bisty: Termination Demokrise	A STATE OF THE PARTY OF THE PAR	- Andrical Palle	pile Afility (detault) Ma, Do Mot Apply
v: 175 F076	word Shat Su	se Publication a	opiles to both	Bection 10.3.2	' = Others	Manual Reserve	Control of the second
	A ADDRESS OF THE PARTY OF THE P	J.		Other Agreement	Other A	Greenistik Sielella	Tig New Assets
for 2.26	■ Gus Daily	Midpoint (default)		Section 14.5			inter whitely
285cm				Choice Of Law	North	Caroline	
ion 6	Buyer Pau	At and After De	heart Balas				
•	(delault)		under Later	Section 14,10	■ Confide	ntiality applies (de	facility)
	O Dellas Co.	Before and At D	olium Data	Confidentiality		ntiality does not a	
Provin	our Manper of a	neets attached:	Arrival A LASH				ाक भ र
							
dend arm(s):			الترج والمستحدد المراجعة				_
	CEPT IN A SAU	1000 hérato have	executed this Ba	ee Contract in dural	Icele.		
NESS WHI	- Market						
TNESS WHI	tibutore/LLC			Sinishilaid &	inter for Man	Dant Inc. of	
dendization; TNESS WHI Tal Gas Died Name	tibuton/LLC			Daniel An	ALMON TO BE A SECOND TO SE	<u>neel location</u> :	
TNESS WHI	movement C			Party Name	*	Heel location)	
TNESS WHI	ributoro/LLC	<u> </u>	**********	Party Name	•		
TNESS WHI	movement C	10		By 234	mile		
TNESS WHI	movement C	7 0		sand impage	Mile-		

General Terms and Conditions Base Contract for Sale and Purchase of Natural Gas

SECTION 1. PURPOSE AND PROCEDURES

1.1. These General Terms and Conditions are intended to facilitate purchase and sale transactions of Gas on a Firm or interruptible basis. "Buyer" refers to the party delivering Gas, The anim agreement between the parties shall be the Contract to deliber in Section 2.7.

The parties have selected either the "Cral Transaction Procedure" or the "Written Transaction Procedure" as indicated on the

Oral Transaction Procedure:

1.2. The parties will use the following Transaction Confirmation procedure. Any Gas purchase and sale transaction may be effectuated in an EDI transmission or telephone conversation with the offer and acceptance constituting the agreement of the parties, shall be considered a "utility" and to have been "eigned". Notwithetending the foregoing sentence, the parties agree that Confirming Party sitial; and the other party may, confirm a telephonic impaction by sending the other party. Transaction: Confirming faculty, EDI-or results agree she electronic means within three Business Days of a transaction congress by this Seiglon 1.2 (Oral parties." Confirming Party stopts its confirming letteries of the transaction confirmation and subjection of Confirming Party. If the Transaction Confirmation any provisions other itse is the transaction of the confirming letteries of the transaction confirmation and subjections of the transaction (i.e., price; quantity, performance obligibles, delivery stont, period of delitions; which modify an supplement the Base Contract or General Terms and Confirming of the transaction and warranties, such provisions shall not be deemed to be accepted purchasing artists agreed to by both parties; provided that the foregoing shall not be deemed to be accepted purchasing and parties.

Written Transaction Procedure:

- 1.2. The perfect will rise the following transaction Confirmation procedure. Should the perties come to an egraphism regarding a Gas purchased seef sale fransaction for a particular Design? Perfod, the Confirming Perfy shall, and the other party risky record that agreement on a Transaction Confirmation and computations such Transaction Confirmation by Modellife; EDI or purply agreement electropic means, to the other party by the close of the Dustness Day following the date of agreement. The particle acknowledge that their rights make the process of the Confirmation of the passage of the Confirmation without objection from the receiving party, as isocided in Section 1.3.
- Describe without objection from the recessing party, as according an section 1.3.

 1.3. If it sending party's transaction Confirmation is marginally different from the receiving party's undergrading of the agreement referred to in Section 1.2. sport receiving party shall notify the sending party via tessiyine, EDI or ritually agreeable described party in the salter of the receiving party in the confirmation of the sending party. The tellure of the receiving party is so notify any sending party. I writing by the Confirmation Challette are any material differences between straight from the transaction for the transaction of the transa
- 1.4. The parties eight stall each party may electrolically second all telephone convertablishes with mispact to this Contract between their respective entitions, without any special or further notice to the other party. Each party shall obtain any heatistary construction and entitle across the such recording. Where the parties have entected the Oral Transaction Procedure in Seption 1.2 of the Base Contract, the parties across not to contest the validity or enforcestity of telephonic recordings entered into in accordance with the requirements of this Base Contract. Flowerer, nothing herein shall be construed as a waiver of any objections of the admissibility of such evidence.

SECTION 2. DEFINITIONS

The terms set forth below shall have the meaning sacribed to them below. Other terms are also defined elsewhere in the Contract and shall have the meanings ascribed to them harein.

- 2.1. "Alternative Dismages" shall mean such damages, expressed in dollars or dollars per MMBtu, as the parties shall agree upon in the Transaction Confirmation, in the event either Seller or Buyer falls to perform a Firm philipation to deliver Ges in the case of Seller or to receive Gas
- 2.2. "Bees Contract" shall mean a contract executed by the parties that incorporates these General Terms and Conditions by reference; that specifies the agreed selections of provisions contained herein; and that sats forth other information required herein and any Special Provisions and addendum(s) as Identified on page one.
- "British thermal unit" or "Blu" shall mean the International BTU, which is also called the Blu (17).

- "Business Day" shall mean any day except Saturday, Sunday or Federal Reserve Bank holidays. 2.4,
- 2.5. "Confirm Deadline" shall mean 5:00 p.m. in the receiving party's time zone on the second Business Day following the Day a Transaction Confirmation is received or, if applicable, on the Business Day agreed to by the parties in the Base Contract; provided, if the Transaction Confirmation is time stamped after 5:00 p.m. in the receiving party's time zone, it shall be deemed received at the
- "Confirming Party" shall mean the party designated in the Base Contract to prepare and forward Transaction Confirmations to the other 2.6. party.
- 2.7. "Contract" shall mean the legalitebridge relationship established by (i) the Base Contract, (ii) any end all binding Transaction Confirmations and (iii) where the parties have been deather than Confirmations that the parties have been confirmed in a EDI transmission or by telephone, but that have not been confirmed in a
- "Contract Price" shall mean the amount expressed in U.S. Dollars per MMBtu to be paid by Buyer to Seller for the purchase of Gue as agreed to by the parties in a traheaction:
- "Contract Quantity" shall make the dignitity of Gas to be delivered and taken as agreed to by the parties in a transaction. 2.9.
- "Cover Standard", as referred to in Section 3.2, shall mean that if them is an unexcused failure to take or deliver any quantity 2.10. "Cover Standard", as referred to in specion 3.2, shall mean may it means an unexcusso minute to tend or server any quartery of Gas pursuant to this Contract, then the performing party shall use commercially reasonable efforts to (i) if Buyer is the performing party, obtain flas, (or an attempte fuel if elected by Buyer, and replacement Gas is not available), or (ii) if Selects the performing party, sell Gas, breather case, are successfully for the delivery of production area, as explicable, opinishing party; the immediately of the Buyer's Gas consumption needs or Sellect Gas sales, highly ments, as applicable; the quartities involved; and the anticipated legish of failure by the nonpetitioning party.
- 2.11. Challif Support Obligation(s) shall mean any obligation(s) to provide or selection credit support for or behalf of a party to the Contract such as an irrevocable exhibitor letter of credit, a margin agreement, it prepayment a security interest in an asset, a performance behalf or other good, and sufficiently of a continuing nature.
- 2.12. "[Jay, Shall indee a period of 24 consecutive hours, coextensive with a "day" as defined by the Becelving Transporter in a particular transportant.
- 2.13. Delicary Pariod shall be the period during which deliveries are to be made as agreed to by the parties in a transaction.
- 2.14. "Delivery Point(s)" shall mean such point(s) example greed to by the parties in a transaction.
- 2.15. EDT shall mean so electronic lights interchange pursuant to an apharment emered this by this parties, specifically relating to the communication of Principation Configuations under this Contract.
- 2.16. "EFF suid-mean the purchase, sale or exchange of mitorial Gas as the "physical" side of an authorise physical transaction involving gas fully contracts. EFP Itial incorporate the meaning and remedies of "Firm", provided that a party's excuse for nonpatienthance of its obligations to deliver or receive Gas will be governed by the rules of the palekint futures attributing regulated under the Springold's Exchange Act.
- 2.17. Frint" shall relate that grant that either party may interrupt to performance without liability only to the either that editorperformance is prevented for reache of Force Majoure; provided, however, that during Force Majoure that menuptions, the party involving figure Majoure may be reache of first interruption after the party involving figure Majoure Transportion and until the distance in deliveres and/or receipts is confirmed by the Transporter.
- 2.18. "Gae" shall mean sary mixture of hydrocarbons and noncombustible gases in a gaseous state contributing prints by of methans.
- 2.19. "Ritislance Startiss" shall mean any fees, parallies, costs or charges (fri cash or in kind) siseased by \$ \tau_i\text{property} failure to satisfy the Emphasioner's balance and/or nominality requirements.
- 2.20. "Interruptible" still mean that eliger party may interrupt its performance at any time for any interior whether or his caused by an event of Porce Majeria, with no Eability accept such interrupting party may be responsible for any initializate Chaiges as set forth in Section 4.3 yeared to its Interruption affective nomination is made to the Transporter against the change in deliveries which receipts is confirmed by Transporter
- 2.21. "Middle" stjelf mean one million. Brillsh themisi units, which is equivalent to one dekathern?
- 2.22. "Month" shall mean the period beginning on the first Day of the calendar month and ending immediately prior to the commencement of the first Day of the next celendar month.
- "Payment Date" shall mean a date, as indicated on the Base Contract, on or before which payment is due Satier for Gas received by Guyer in the previous Month.
- 2.24. "Receiving Transporter" shall mean the Transporter receiving Gas at a Delivery Point, or absent such receiving Transporter, the Transporter delivering Gas at a Delivery Point.
- 2.25. "Scheduled Gas" shall mean the quantity of Gas confirmed by Transporter(a) for movement, transportation or management.
- 2.25. "Spot Price" as referred to in Section 3.2 shall mean the price field in the publication indicated on the Base Contract, under the listing applicable to the geographic location closest in proximity to the Delivery Point(s) for the relevant Day; provided, if there is no single price published for such location for such Day, but there is published a range of prices, then the Spot Price shall be the average

of such high and low prices. If no price or range of prices is published for such Day, then the Spot Price shall be the average of the following: (i) the price (determined as stated above) for the first Day for which a price or range of prices is published that next precedes the relevant Day; and (ii) the price (determined as stated above) for the first Day for which a price or range of prices is published that next follows the relevant Day.

- "Transaction Confirmation" shall mean a document, similar to the form of Echibit A, setting farth the terms of a transaction formed pursuant to Section 1 for a particular Delivery Pariod.
- 2.28. "Termination Option" shall mean the option of either party to terminate a transaction in the event that the other party fails to perform a Firm obligation to deliver Gas in the case of Shight or to receive Gas in the case of Buyer for a designated number of days during a period as specified on the applicable transaction Configuration.
- 2.29. "Transporter(e)" shall regish all Gast guidering or pipeline companies, or food distribution companies, acting in the capacity of a transporter, transporting Gas for Selfer or Buyer upstright or clownstream, respectively, of the Dollvery Point pursuant to a particular framsporter.

PERFORMANCE OBLIGATION

3.1. Seller agrees to sell and deliver, and Buyer agrees to receive and punchase, the Contract Quantity for a perticular transaction in accordance with the terms of the Contract. Sales and punchases will be on a Firm or interruptible basis, as agreed to by the parties in a

The parties have selected either the "Cover Standard" or the "Spot Price Standard" as indicated on the Seas Contract. Cover Stringerd:

Gover Standard:

3.2. The sole and acclusive remedy of the parties in the event of a breach of a Birm obligation to deliver or retains Gas shall be recovery of the following: (i) in the event of a breach lip. Belier on any Enyle), payment by Seller of Blayer, in an amient squal to the positive difference, it any, histored the purchase price trait by Buyer utilizing the Coyer Standard and the Contract Price in an amient squal to the positive difference, it any, better or price the contract Quantity and the country delivered by Seller for such Qay(a); or ((ii) in the event of Seller's by Buyer or any Day(a), payment by Buyer to Seller's the amount equal to the positive difference, it any, bettered the Contract Online it is the seller for the result of the positive difference, it any, bettered the Contract Online it is the positive of the contract Online it is the contract of the seller in the seller of the seller in the seller of the seller in the contract of the contract Online it is the contract of the seller in the seller of the seller in the contract of the contract of the seller in the se

Spot Price: Sienders:

- 3.2. This acts and exclusive remedy of the parties in the event of a breach of a time obligation to before or nation this shall be recovery of the following: (i) in the event of a breach by Seller on any Day(s), payment by Seller in the Seller in an emperiting and the actual strainty delivered by Seller in the Seller in th
- Notwithstanding Section 3.2, the parties may agree to Alternative Damages in a Transaction Confirmation obtained in writing by both parties.
- 3.4. In addition to Sections 3.2 and 3.3, the parties mily provide for a Tegralmation Option in a Transaction Confirmation executed in writing by both parties. The Transaction Confirmation containing the Termination Option will designate the length of nonperformance riggering the Termination Option and the procedures for exercise thereof, how damages for nonperformance will be compensated, and how liquidation costs will be calculated.

SECTION 4. TRANSPORTATION. NOMINATIONS, AND IMBALANCES

- Seller shall have the sole responsibility for transporting the Gas to the Delivery Point(s). Buyer shall have the sole responsibility for transporting the Gas from the Delivery Point's).
- 4.2. The parties shall coordinate their nomination activities, giving sufficient time to meet the deadlines of the affected Transporter(s). Each party shall give the other party timely prior Notice, auticient to meet the requirements of all Transporter(s) involved in the transaction, of the quantities of Gae to be delivered and purchased each Day. Should either party become aware that actual deliveries at the Delivery Point(s) are greater or lesser than the Scheduled Gae, such party shall promptly notify the other party.

4.3. The priries shall use commercially reasonable efforts to swold imposition of any imbalance Charges. If Buyer or Seller receives an invoke from a Transporter that includes imbalance Charges, the parties shall determine the validity as well as the cause of such imbalance Charges. If the imbalance Charges were incurred as a result of Buyer's receipt of quantities of Gas greater than or less than the Scheduled Gas, incurred as a result of Seller's delivery of quartities of Gas greater than or less than the Scheduled Gas, then Seller shall pay for such imbalance Charges were Charges or reimburse Buyer for such imbalance Charges paid by Buyer.

QUALITY AND MEKSUREMENT

All Ges delivered by Seller shall meet the principle, quality and heat content requirements of the Receiving Transporter. The unit of quantity measurement for purposes of this Contract shall be form MASte dry. Measurement of Gas quantities hereunder shall be in accordance with the

SECTION 6.

The parties have selected either "Buyer Pays At and Alter Delivery Point" or "Seller Pays Before and At Delivery Point" as indicated on the Base Contract.

Buyer Paye At and After Delivery Point:

Selier shall pay or cause to be paid all taxes, take, levies, penalties, licenses or charges imposed by any government authority (Taxes') on or with respect to the Gas, prior to the Delivery Point(s). Buyer shall pay or cause to be paid all Taxes on or with respect to the Gas, at the Delivery Point(s) and all Taxes after the Delivery Point(s). If a party is required to result or pay Taxes that are the different party is perturbed, all party is perturbed to result or pay Taxes. Any party entitled to an examplion from any stock Taxes, or charges shall surplish the other party any necessary documentation thereof.

Baller Pagis: Ballore and At Dallvery Point:

Sellor shall, pay(or cariste to see paid all traces, force, teviate, partitions, it canses, or charges interpeed by any doctors and sufficiently of partitions and all families at the Delivery Point(s). Buyer shall play or cause to be be in a with respect to she Gais after the Delivery Point(s), and all families at the Delivery Point(s). Buyer shall play or cause to be be in a with respect to she Gais after the Delivery Point(s), (if a party is required to remain or pay Taxass that are all a other party is pay assumed the party responsibility from any such Taxas, any party entitled is any assumption any such Taxas, any party entitled in any assumption.

SECTION 7. BILLING, PAYMENT, AND AUDIT

- 7.1. Setter shall justice Buyer for Gas delivered spit received in the preceding Month and for any object applicable integrate, providing applicable in the property of the set of the set
- 7.2. Slayer strill retrill the amount due (inder Section 7.7. In the meaner specified in the Base Contract; in territalistic funds, on or before fire Payment Date or 10 Says after accept of the invoice by Buyer; provided that if the Payment Date is not a Sustance Day, payment is due on the next Bigainess Day killowing that bline. In the event any payments are due Buyer ferrounder, payment to Boyer shall be made inaccontance visit this Section 7.2.
- 7.3. In the every payments become the pursuant to Sections 3.2 or 3.3, the performing gardy may submit the knotice to the nonperforming party for an accelerated payment eating forth; the bests upon which the loveload empurity was calculated. Payment from the nonperforming party will be due tive Business Day's after receipt of impice.
- 7.4. If the invoicely party, in good fails, disputes the impurit of any such invoice or any partitioned, such invited party will party such amount as a deceptible to be correct, provided, increases, if the invoiced party disputes the amount due, it must be provided, supporting documentation may pursue any remedy available at law or in acuty to entirior bursuant to this Section.
- 7.5. If the involced perfectals to remit the full amount payable when due, interest on the unpaid performabilishare from the date of payment at a rate equal to the lower of () the then disclive prime rate of interest published until Microsy Retest by The Well Street Journal, plus two percent for arrunn or (i) the maximum applicable levels interest rate.
- 7.6. A party shall have the right, at its own expense, upon reasonable Notice and at reasonable times, it blacking with facilit and to obtain copies of the other party only to the extent reasonably necessary to vally not be extent reasonably necessary to vally not be extent reasonably necessary to vally not be sufficient or y statement, or computation made under the Contract. This right to examine, such, and to obtain copies shall not be sucliable with respect to proprietry information not directly relevant to transctions under this Contract. All invoices and billings shall be conclusively presumed final and accurate and all associated claims for under to everpayments shall be deemed waived unless such invoices or billings are objected to in writing, with adequate explanation endor documentation, within two years after the Month of Gas delivery. All retractive adjustments under Section 7 shall be paid in full by the party owing payment within 30 Days of Notice and substantiation of such inaccuracy.
- Unless the parties have elected on the Base Contract not to make this Section 7.7 applicable to this Contract, the parties shall not of the new payment of the net amount to the other party in accordance with Section 7; provided that no payment required to the net amount to the other party in accordance with Section 7; provided that no payment required to the net amount to the other party in accordance with Section 7; provided that no payment required to the net amount to the other party in accordance with Section 7; provided that no payment required to the net amount to the other party in accordance with Section 7; provided that no payment required to the net amount to the other party in accordance with Section 7; provided that no payment required to the net amount to the other party in accordance with Section 7; provided that no payment required to the net amount to the other party in accordance with Section 7; provided that no payment required to the net amount to the other party in accordance with Section 7; provided that no payment required to the net amount to the other party in accordance with Section 7; provided that no payment required to the net amount to the other party in accordance with Section 7; provided that no payment required to the net amount to the other party in accordance with Section 7; provided that no payment required to the net amount to the other party in accordance with Section 7; provided that no payment required to the new made pursuant to the terms of any Credit Support Obligation or pursuant to Section 7.3 shall be subject to netting under this Section. If the parties have executed a separate netting agreement, the terms and conditions therein shall prevail to the extent inconstatent

SECTION 8. TITLE, WARRANTY, AND INDEMINITY

- 8.1. Unless otherwise specifically agreed, title to the Gas shall pass from Seller to Buyer at the Delivery Point(s). Salier shall have responsibility for and assume any liability with respect to the Gas prior to its delivery to Buyer at the specified Delivery Point(s). Buyer shall have responsibility for and any liability with respect to said Gas after its delivery to Buyer at the Delivery Point(s).
- 8.2. Seller warrants that it will have the light to convey and will transfer good and merchanizable title to all Gas sold hereunder and delivered by it to Buyer, free and clear of all flens, encumbrances, and clears. EXCEPT AS PROVIDED IN THIS SECTION 8.2 AND IN SECTION 14.8, ALL OTHER WARRANTIES, EXPRESS OR IMPLIED, INCLIDING ANY WARRANTY OF MERCHANTABILITY OR OF PITNESS FOR ANY PARTICULAR PURPOSE, ARE DISCLAMED.
- 5.3. Seller agrees to indemnity. Buyer abid knive it harmless from all losses, itabilities or claims including reasonable attorneys fees and costs of court ("Claims"), from any and all persons, arising from or out of claims of title, personal injury or property damage from said Gas or other charges thereos, which stach before title passes to Buyer. Buyer agrees to indemnity Seller and save it harmless from all Claims, from any and all persons, stising from or out of claims regarding payment, personal lightly or property damage from said Gas or other charges thereon which effect pleases to thisper.
- 8.4. Nobilitetanding the other provisions of star Section 8, as between Seller and Suyer, Seller will be liable for all Claims to the extent that such arise from the faiture of Gas delivered by Seller to meet the quality requirements of Section 6.

SECTION 9.

- 9.1. All Transaction Confirmations, invoices; payments and office convinuincellons made pursuant to the Base Contract (Notices') shall be made to the addresses specified by writing by the respective parties from time to time.

 9.2. All lightces required hereunder may be sent by tacsimile or mutually acceptable electronic making a nationally recognized overnight courier service. Test class made or hand delivered:
- 9.3. Notice shall be given when received on a Büsiness Day by the addressee. In the absence of proof of the actual receipt date, the following presumptions will apply. Notices sent by facetimile shall be deemed to have been reached upon the shipting perty's receipt of its actimile machines confination of successful transmission. If the day on which such actually is registed is not a Business Day, there's been facetimide shall be defined to have been received on the notification of such as a Business Day, there's the been received on the notifications of a Business Day, and a state of the sent state ·after milling. SECTION 10. FINANCIAL RESPONSIBILITY

- 10.1. If either party (X) has reasonable grounds for insecurity regarding the performance of any obligation until this Contract (whether or not titled, by the other party (Y) (including without limitalism, the observation of a material burnes in the credit of the contract of performance. Adequate Assurance of Performance. Adequate Assurance of Performance what mean sufficiently in the form, amount sold for the term reasonably acceptable to X including, but not improve the country interest in an asset or a performance bond or gueranty stocking the leavet of any such
- 10.2. In the event (sectron "Event of Default") either party (the "Defaulting Party") or its guaranter-situal; (i) make an exhigement or any general arrangement for the banefit of creditors; (ii) file a petition or otherwise commange, surjuptize, or applicate in the commange commange, surjuptize, or application in the commangement of a phage against it; (iii) spirotes and party or similar interest of commanders applications and they sife due; (ii) have a receiver, provisional liquidator, conservator, custodian, trustale or other similar commanders possible surjuptions are relating to the Contract; (vii) fall to give: Adequate Assulance or Performance under-Section 10x significations (vii) party in give party or (viii) not have paid any amount that the other party the other party in give best one second Sustainess Day following written Notice that such payment is due; then the other party (the "Noti-Defaulting Party") which have the right, at its sole election, to immediately withhold and/or suspend deliveries or payments upon Notice and/or to terminate available the transactions under the Contract, in the manner provided in Section 10.3, in addition to any and all other remettes available hareunder.
- 10.3. If an Event of Default has occurred and is continuing, the Non-Defaulting Party shall have the right, by Notice to the Defaulting Party, to designate a Day, no earlier than the Day such Notice is given and no later than 20 Days after such Notice is given, as an early termination data (the "Early Termination Date") for the figuidation and termination pursuant to Section 19.3.1 of all transactions under the Contract, each a "Terminated Transaction". On the Early Termination Date, all transactions will terminate, other their those transactions, if any, that may not be liquidated and terminated under applicable law or that are, in the reasonable opinion of the Non-Defaulting Party, commercially imprecisable to liquidate and terminate ("Excluded Transactions"), which Excluded Transactions must be liquidated and terminated as soon thereafter as is reasonably practicable, and upon termination shall be a Terminated Transaction and be valued consistent with Section 10.3.1 below. With respect to each Excluded Transaction, its actual termination date shall be the Early Termination Date for purposes of Section 10.3.1.

The parties have selected either "Early Termination Damages Apply" or "Early Termination Damages Do Not Apply" as indicated on the Base Contract.

Early Termination Demages Apply:

10.3.1. As of the Early Termination Date, the Non-Defaulting Party shall determine, in good falls and in a commercially reasonable maturer. (I) the amount owed Autotine and Excluded Termination party with respect to all Gas delivered and received between the parties under Terminated Transactions and Excluded Termination on and before the Early Termination Date and all other applicable charges relating to such deliveries and receipts (Including without limitation any amounts owed under Section 2.2), for which of each Terminated Transaction. The Nort-Defaulting Party shall (A) liquidate and accelerate each Terminated Transaction at its such Market Value, so that each amount equal to the biliferance between such Market Value and the Contract Value, as defined below, of Contract Value and to the Selliat Rive opposite is the case; and (y) where appropriate, discount each emount then due under clause (b) above to present value in a commercially reasonable manner as of the Early Termination Date (to trice account of the period between the date of Rouldation and the date of which such amount would have otherwise been due pursuant to the relevant Terminated Transactions).

Terminated Transactions).

For purposes of this Section 10.3.1, "Contract Value" means the amount of Gas remaining to be delivered or purchased under a transaction multiplied by the Contract Price, and "Market Value" means the amount of Gas remaining to be delivered or purchased under a transaction multiplied by the Contract Price, and "Market Value" means the amount of Gas remaining to be delivered or purchased under a transaction multiplied by the market price for a similar transaction at the Delivery Point determined by the Blon-Defaulting Party men) containing Party may or all of the settlement prices of NYMEX Gas futures contracts, quotations from leading dealers in the contracts or physical gas fixeding markets; similar solver or purchases and any other bons fide third-party offers, all districts of the length of the time unit districts of the solver of a party shall not be required to enter into a indiscement transaction; in the corder to determine the Market Value. Any common of the term of a transaction to which period are into bound as a the Earty Market Values. For the dybidence of doubt, any option purposes to which one party has the right to apparent the term of a figure and Market Values. For the dybidence of doubt, any option purposes to which one party has the right to apparently the term of a figure and Market Values. The rate of interest used in calculating net present value shall be delarminated to the Non-Defaulting Party in a common cluster reasonable manner.

Early Termination Demanes Do Not Apply:

10.3.1. As of the Early Termination Date, the Non-Defaulting Party shall determine, in good faith and in a commercially respectable then her, the entrumt owed (Wiether or not trian due) by each pility with respect to all Gas delivated and received between applicable citizens matching to such delivates and Excluded Transactions on, and before the Early Termination Path and all other applicable citizens matching to such delivates and receipts including without faithful and simple mounts own Junior Section 3.2), for which payment has dot yet bean made by the party that owns such payment under this Contract.

The parties have delected atther "Other Agreement Setoffs Apply" or "Other Agreement Setoffs Do Not Apply" se indicated

Other Adisement Setulle apply:

10.3.2. The Nets-Defaulting Party shall net or aggregate, as appropriate, any and all amounts oviling between the parties under Section 10.3.1, so that all such amounts are netted or significated to a single squidated amount payable by one parties ited other (the Settlemann Ansount). At its sole opition and without prior Notice to the Defaultion Party, the Notice parties are Not Settlemann Ansount Ansount owned to the Non-Defaulting Party against any mergin of other collegetion relating to the Contract; or (ii) any Net Settlemann Ansount payable to the Defaulting Party significant any amount(s) payable by the Defaulting Party to the Non-Defaulting Party under any other agreement or arrangement between the parties.

Other Kareement Betoffs Do Not Apply:

10.3.2. The Non-Diffecting Party shall not or aggregate, as appropriate, any and all amounts owing between the parties under Section 10.3.1, so that all stich amounts are noted or aggregated to a single liquidated amount payable by one party in the "Not Settlement Amount). At its sole option and without prior Notice to the Defaulting Party, the Non-Defaulting Party had settlement Amount owned to the Non-Defaulting Party against any margin or other collaterst held by it in connection with any Cracit Support Obligation relating to the Contract.

- 10.3.3. If any obligation that is to be included in any notting, aggregation or seloff pursuant to Section 10.3.2 is unascertained, the Non-Defaulting Party may in good feith estimate that obligation and not, aggregate or setoff, as applicable, in respect of the estimate, subject to the Non-Defaulting Party accounting to the Defaulting Party when the obligation is ascertained. Any amount not then the which is included in any notting, aggregation or setoff pursuant to Section 10.3.2 shall be discounted to not present value in a commercially reasonable menner determined by the Non-Defaulting Party.
- 10.4. As soon as practicable after a liquidation, Notice shall be given by the Non-Defaulting Party to the Defaulting Party of the Net Settlement Amount, and whether the Notice shall include a written statement explaining in reasonable detail the calculation of such amount, provided that failure to give such Notice shall not affect the validity or enforceability of the liquidation or give rise to any claim by the Defaulting Party against the Non-Defaulting Party. The Net Settlement Amount shall be paid by the close of business on the second Business Day following such Notice, which date shall not be earlier than the Early Termination Data. Interest on any unpaid portion of the Not Settlement Amount shall accuse from the date due until the

date of payment at a rate equal to the lower of (i) the then-effective prime rate of interest published under "Money Rates" by The Wall Street Journal, plus two percent per annum; or (ii) the medimum applicable lawful interest rate.

- 10.5. The parties agree that the transactions hereunder constitute a "forward contract" within the meaning of the United States Bankruptcy Code and that Buyer and Salier are each "forward contract merchants" within the meaning of the United States Bankruptcy
- 10.6. The Non-Defaulting Party's remediate under this Section 10 are the sole and exclusive remedies of the Non-Defaulting Party with respect to the occurrence of any Early Namination Date. Such party reserves to itself all other rights, setoffs, counterclaims and other defances that it is or may be aristed the initialing from the Contract.
- 10.7. With respect to this Section 10, if the parties have executed a separate netting agreement with close-out netting provisions, the terms and conditions therein shall prevail to the extent incornectant herewith.

SECTION 11. FORCE MAJIBURE

- 11.1. Except with regard to a party's obligation to make payment(s) due under Section 7, Section 10.4, and imbalance Charges under Section 4, neither party shall be liably to the objector failure to perform a First obligation, to the extent such failure was caused by Force suspension, as further defined in Section 14.2.
- 11.2. Force Majaure shall include, but not be limited to, the following: (i) physical events such as acts of God, landsities, lightning, carringtains, fires, storings or storm warnings, such as hurricants, which resolt in evenuation of the affected area; floors, weathouts, explosine, invasions or acceptance of the explosions, invasions or explosions and areas to the companion of the affected area; floors and an artist programment or fines of plus. (ii) washing in redsoling our statistics of plus of plus (iii) washing an and/or storage by Transporters; (iv) each of wells or state of plus; (iii) alternation and/or storage by Transporters; (iv) each of others such as at lines; (iii) alternation and/or storage by Transporters; (iv) each of others such as at lines; (iii) alternation or distributions; (iii) and iii) and or other industrial distributions; (iii) and iii) are countered or of the industrial actions; (iii) are governmental state or a count or other industrial actions; (iii) are storage in a countered in other to resume performance.
- 11.3. Neither party shall be entitled to the bonest of the provisions of flores Majeury to the extent performance in affected by any or all of the following circumstances: (i) the curtalinant; of intermediate performance in affected by any or transposition is also curtalined; (ii) the party claiming excess falled to remedy the obtdition and to results are performing of such coverents, or obligations with reasonable depetor; or (iii) economic bandetile, to include, without irritation; Spiler's ability by the dess at a higher of more advantageous price than the Contract Price, one regulatory agency deptember, in white or its party to provide a party to provide a party deptember of costs of such as a party of the cost of such as a party of the costs o
- 11.4. Notwithstanding shything to the contrary herein the parties sories that the settlement of strikes, lockouts or other industrial disturbances shall be within the sole discretion of the parties periencing such disturbance.
- 11.5. The party whose biriormance is prevented by Force Majeure must provide holics to the other party. Initial holice may be given grafty, povered, writish holice with reasonably full particulars of the event or occurrence is required as reliably possible. Upon providing writing holice of Force Majeure to the philip party, the affected party will be reliaved of the charge to have the party of Gas, as applicable, to the extent and far, the duration of Force Majeure, and risk many party shall be described to have talked in such obligations to the other clusters occurrence or event.
- 11.6. Natvithstanding Sections 11.2 and 11.3, the parties may agree to alternative Force Majoure provisions in a Francection Confirmation specuted in willing by both parties.

SECTION 12. TERM

This Contract may be terminated on 30 Day's written highire, but shall remain in effect until the expiration of the listest Delivery, Pariod of any transaction(a). The rights of either party pursuant to Section 7.6 and Section 10, the obligations to make payment hereinflar and the obligation of either party to incleasily the other, pursuant herein shall survive the termination of the Base Contract of any transaction.

SECTION 13. LIMITATIONS

FOR BREACH OF ANY PROVISION FOR WHICH AN EXPRESS REMEDY OR MEASURE OF DAMAGES IS PROVIDED, SUCH EXPRESS REMEDY OR MEASURE OF DAMAGES SHALL BE THE SOLE AND EXCLUSIVE REMEDY. A PARTY'S LIABILITY HERELANDER SHALL BE LIMITED AS SET FORTH IN SUCH PROVISION, AND ALL OTHER REMEDIES OR DAMAGES AT LAW OR IN EQUITY ARE WANED, IF NO REMEDY OR MEASURE OF DAMAGES IS EXPRESSLY PROVIDED HEREIN OR IN A TRANSACTION, A PARTY'S LIABILITY SHALL BE LIMITED TO DIRECT ACTUAL DAMAGES ONLY. SUCH DIRECT ACTUAL DAMAGES SHALL BE THE SOLE AND EXCLUSIVE REMEDY, AND ALL OTHER REMEDIES OR DAMAGES AT LAW OR IN EQUITY ARE WANED. UNLESS EXPRESSLY HERBIN PROVIDED, NEITHER PARTY SHALL BE LIABLE FOR CONSEQUENTIAL, INCIDENTAL, PUNITIME, EXEMPLARY OR INDIRECT DAMAGES, LOST PROFITS OR OTHER BUSINESS INTERRUPTION DAMAGES, BY STATUTE, IN TORT OR CONTRACT, UNDER ANY NOEMNITY PROMISION OR OTHERWISE. IT IS THE INTENT OF THE PARTIES THAT THE LIMITATIONS HERBIN IMPOSED ON THE MEASURE OF DAMAGES BE WITHOUT REGARD TO THE CAUSE OR CAUSES RELATED THERETO, INCLUDING THE MEGLIGENCE OF ANY PARTY, WHETHER SUCH NEGLIGENCE SE SOLE, JOINT OR CONCURRENT, OR ACTIVE OR PASSING.

Copyright © 2002 North American Energy Standards Board, Inc. All Rights Reserved

NAESS Standard 5.3.1 April 19, 2002 TO THE EXTENT ANY DAMAGES REQUIRED TO BE PAID HERELAIDER ARE LIQUIDATED, THE PARTIES ACKNOWLEDGE THAT THE DAMAGES ARE DIFFICULT OR IMPOSSIBLE TO DETERMINE, OR OTHERWISE OBTAINING AN ADEQUATE REMEDY IS INCONVENIENT AND THE DAMAGES CALCULATED HEREUNDER CONSTITUTE A REASONABLE APPROXIMATION OF THE HAVEN OR LOSS.

SECTION 14. MISCELLANEOUS

- 14.1. This Contract shall be binding upon and inure to the benefit of the successors, assigns, personal representatives, and heirs of the respective parties hereto, and the coverants, conditions, rights and obligations of this Contract shall not for the full term of this Contract. No assignment of this Contract, in whole or in part, will be made without the prior written consent of the non-assigning party (and shall not relieve the assigning party from liability herecitating, whith consent will not be unreasonably withheld or detayed; provided, either party may (i) transier, sell, placing, encumber, or assign this Contract of the accounts, revenues, or proceeds hereof in connection with any financing or other financial amengements, or (it transier its interest to any parent or affiliable by assignment, marger or otherwise without the prior approval of the other party. Upon any such assignment, transfer and assumption, the transferor shall remain principally liable for and shall not be relieved of or discharged from any obligations the currents.
- 14.2. If any provision in this Contract is determined to be invalid, void or unenforceable by any court having jurisdiction, such determination shall not invalidate, void, or make unenforceable sing other provision, agreement or coverant of the Contract.
- 14.3. No water of any breach of this Contract shall be held to be a weiver of any other or subsequent breach.
- 14.4. This Contract sets forth all understandings between the parties respecting each transaction subject hereto, and any prior contracts, understandings and representations, whether orel or written, relating to such transactions are merget life (and superseded by this Contract and only effective transaction(s). This Contract may be amended only by a writing executed by both parties.
- 14.5. This interpretation and performance of this Christochasti be governed by the laws of the jurisdiction as indicated in Base Contract, stroitising, however, any conflict of laws rule which would apply the lay of enotine pricing in the law of the laws rule which would apply the lay of enotine pricing in the law of the laws rule which would apply the lay of enoting the law of the laws rule which would apply the lay of enoting the laws rule which would apply the lay of enoting the laws rule which would apply the lay of enoting the laws rule which would apply the lay of enoting the laws rule which would apply the lay of enoting the laws rule which would apply the lay of enoting the laws rule which would apply the lay of enoting the laws rule which would apply the lay of enoting the laws rule which would apply the lay of enoting the laws rule which would apply the lay of enoting the laws rule which would apply the lay of enoting the laws rule which would apply the lay of enoting the laws rule which would apply the lay of enoting the laws rule which would apply the lay of enoting the laws rule which would apply the lay of enoting the lay of enoting the laws rule which would apply the lay of enoting the laws rule which would be ruled to the laws rule which would be ruled to the laws ruled t
- 14.6. This Contract strifted provisions herein will be subject to all applicable and valid statutes, rules, orders and taggitations of any governmental authority family; jurisdiction over the parties, their judities, or God supply, that Contract or tensecubities any phonistic thereof.
- 14.7. There is no third party beneficiary for this Contract.
- 14.8. Each barry to \$15 Contract represents and warrants that it has full and complete authority to smill this and perform this Contract. Each person this executes this Contract on behalf of either party represents and warrants full it has full and complete authority to do so and that such party-will be bound floreby.
- 14.9. The heatings out subheatings contained in this Contract are used solely for constitutionally do not constitute apent of this Contract between the payment of the format the position of the Contract.
- Contracts between the period and shall not by used to consider in interpret the provisions of this Contract.

 14.10. Utilists the period interest on the Base Contract not to make this Section 14.10 applicable or the Contract makes of the period of the pe

14.11 The parties may agree to dispute resolution probedures in Special Provisions attached to the Base Contract or in a Transaction Confirmation executed in writing by both parties:

DISCLABLER: The purposes of the Contract and is bothed tode, and interconstruction more define the same of contracts of purchase and eals of related. Purber, NAESS does not mended the use of the Contract party party. NAESS disclasses with excludes, and any uses of the Contract account and excludes, and any uses of this contract or any party defined, consistions or representations. Express or empley, oral or written, with respect to this contract or any party defined, including any and all explete warranties or contract or information and the party and all explete warranties or contract in these sor suitability for any party class. Purposes (whether or not naise issued, as related to know, has been advised, or in otherwise in fact analise of any elicity purpose, whether or not naise is year, as related to contract also acress that under no creates will naise be liable for any direct, special, incrental, expended, purpose, purpose of contract also acress that under no creates will naise be liable for any direct, special, incrental, expended, purpose, purpose or consequental damages arising out of any use of the contract.

Letterheid/Log	(a			Dele:Transaction	Confirmation #:	
This Transaction Confirmation is terms of this Transaction Confirm specified in the Base Confract.	publicate b	he Base Contract Inding Unless die	ot between Sell pulsed in writing	er and Buyar d within 2 Busi	eled tess Days of rece	of unless otherwise
SELLER: National Gas Distributors P. O. Sox 85051, 1314 D Resid	t) Road		BUYER:			
Alfri Paul Lewiso Phone: 910-484-4184			Alin: Phone:			
Base Cardidot No. Transporter: Transporter: Transporter:			Transporte	act No.t		
Contract Whee: S. Wilder	200	7 Table 2 Tabl	Transporter	Contract Num	ber:	
Deliviny Period: Bladin:	- 1 3 · · · · · · · · · · · · · · · · · ·		End:			
Performation Obligation and Con	ntracti@uer					7. 20
D ELD With Mark Color (Mark Color Co	•	\ XA .	ble Querifig); (Studday Mint) (Studday Mint)	pun (Alpas.	ole: Silkijimiskiny
		中央成為 3	ection 4.2. 30 e 3 Seller	action of		
Delivery Trajint(d): (If a psolicity polosity limins, list a spe	ciffo geogra	phic ark pipelin	n location:			
Special Godelstones						
				# ¥ 		ed, ed
offer: National Gas Distributors	ue:					7 / A / A
y:		<u> </u>	Buyer:		1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	1 (2) for
No:President			Tille:		···	

EXHIBIT B-2

FW: Strip

Page 1 of T

From: Miller, Bob [BobMiller@smithfieldfoods.com]

Sent: Friday, April 02, 2004 8:57 AM

To: Whelan, Casey Subject: FW: Strip

FYI!
Bob Miller
Smithfield Foods, Inc
111 Commerce Street - PO Box 9002
Smithfield, VA 23430 - Zip 23431
Phone: 757-357-8128

Fax: 757-357-8136

---Original Message----

From: HPCBBALL@aol.com [SMTP:HPCBBALL@aol.com]

Sent: Friday, April 02, 2004 9:21 AM bobmiller@smithfieldfoods.com

Subject: Strip

Bob:

Per our telephone conversation this morning:

- 1) I have secured as part of our "general supply needs" \$5.70 gas for the next 12 mos. This can be "colored" Smithfield Foods in essence capping your price of gas in the event that we run out of nat gas, oil, and all the other petroleum products.
- 2) I am aggressively pushing NCNG to lower your neg t-rate for the long term deal.

Paul Lawing, Jr National Gas Distributors 910-484-9164 office 910-988-2320 cell 910-484-9109 fax

EXHIBIT B-3

From:

HPCBBALL@aol.com

Sent:

To:

Friday, February 25, 2005 7:08 PM BobMiller@smithfleidfoods.com cwhelan@usenergyservices.com
Re: Natural Gas - North Carolina Plants

Cc: Subject:

Bob,

Per our telephone conversation, I have \$5.75 gas plus basis for the summer (Apr-Oct) and I am still trying to nail down our Nov -Mar piece. We can color that gas Smithfield if you wish.

I will be in and out on Monday and Tuesday of next week. Let's talk on Wednesday.

Paul Lawing, Jr National Gas Distributors Phone 910-484-9164 Fax 910-484-9109 Cell 910-988-2320

Niller, Bob

From: HPCBBALL@aol.com

Friday, August 12, 2005 2:07 PM To:

bobmillar@smithfieldfoods.com

Subject: Winter

Bob,

To confirm our recent telephone conversation - this is to confirm a \$6.60 cap for all volumes for all the NC plants.

Thunksi

Paul Lawing, Jr. 910-484-9154 Office 910-484-9109 Fex 910-988-2320 Call

Through march '06

EXHIBIT C

whether or not such interest or lien is provided by or is dependent on a statute and whether or not such interest or lien is made fully effective by statute;

- (53A) The term "stockbroker" means person-
- (A) with respect to which there is a customer, as defined in section 741 of this title; and
- (B) that is engaged in the business of effecting transactions in securities-
- (i) for the account of others; or
- (ii) with members of the general public, from or for such person's own account;
- (53B) The term "swap agreement" means (A) means— (i) any agreement, including the terms and conditions incorporated by reference in such agreement, which is-(A) an agreement (including terms and conditions incorporated by reference therein) which is a rate swapagreement, basis swap, forward rate agreement, commodity swap, interest rate option, forward foreign exchangeagreement, spot foreign exchange agreement, rate cap agreement, rate floor agreement, rate collar agreement, currency swap agreement]) an interest rate swap, option, future, or forward agreement, including a rate floor, rate cap, rate collar, cross-currency rate swap-agreement, currency option, any other similar agreement (including, and basis swap: (II) a spot, same day-tomorrow, tomorrow-next, forward, or other foreign exchange or precious metals agreement:
 - (III) a currency swap, option, future, or forward agreement;
 - (IV) an equity index or equity swap, option, future, or forward agreement;
 - (V) a debt index or debt swap, option, future, or forward agreement;
 - (VI) a total return, credit spread or credit swap, option, future, or forward agreement;
 - (VII) a commodity index or a commodity swap, option, future, or forward agreement; or
 - (VIII) a weather swap, weather derivative, or weather option;
- (ii) any agreement or transaction that is similar to any other agreement or transaction referred to in this paragraph and that--
- (I) is of a type that has been, is presently, or in the future becomes, the subject of recurrent dealings in the swap markets (including terms and conditions incorporated by reference therein); and
- (II) is a forward, swap, future, or option on one or more rates, currencies, commodities, equity securities, or other equity instruments, debt securities or other debt instruments, quantitative measures associated with an occurrence, extent of an occurrence, or contingency associated with a financial, commercial, or economic consequence, or economic or financial indices or measures of economic or financial risk or value;
 - (iii) any combination of agreements or transactions referred to in this subparagraph;
- (iv) any option to enter into any of the foregoing);an agreement or transaction referred to in this subparagraph:
 - (B) any combination of the foregoing; or
 - (C) a master agreement for any of the foregoing together with all supplements;
- (v) a master agreement that provides for an agreement or transaction referred to in clause (i), (iii), (iii), or (iv), together with all supplements to any such master agreement, and without regard to whether the master agreement contains an agreement or transaction that is not a swap agreement under this paragraph, except that the master agreement shall be considered to be a swap agreement under this paragraph only with respect to each agreement or transaction under the master agreement that is referred to in clause (i), (ii), (iii), or (iv); or
- (vi) any security agreement or arrangement or other credit enhancement related to any agreements or transactions referred to in clause (i) through (v), including any guarantee or reimbursement obligation by or to a swap participant or financial participant in connection with any agreement or transaction referred to in any such clause, but not to exceed the damages in connection with any such agreement or transaction, measured in accordance with section 562; and
- (B) is applicable for purposes of this title only, and shall not be construed or applied so as to challenge or affect the characterization, definition, or treatment of any swap agreement under any other statute, regulation, or rule, including the Securities Act of 1933, the Securities Exchange Act of 1934, the Public Utility Holding Company Act of 1935, the Trust Indenture Act of 1939, the Investment Company Act of 1940, the Investment

Advisers Act of 1940, the Securities Investor Protection Act of 1970, the Commodity Exchange Act, the Gramm-Leach-Bliley Act, and the Legal Certainty for Bank Products Act of 2000;

- (53C) The term "swap participant" means an entity that, at any time before the filing of the petition, has an outstanding swap agreement with the debtor;
- (56A) The term "term overriding royalty" means an interest in liquid or gaseous hydrocarbons in place or to be produced from particular real property that entitles the owner thereof to a share of production, or the value thereof, for a term limited by time, quantity, or value realized;
- (53D) The term "timeshare plan" means and shall include that interest purchased in any arrangement, plan, scheme, or similar device, but not including exchange programs, whether by membership, agreement, tenancy in common, sale, lease, deed, rental agreement, license, right to use agreement, or by any other means, whereby a purchaser, in exchange for consideration, receives a right to use accommodations, facilities, or recreational sites, whether improved or unimproved, for a specific period of time less than a full year during any given year, but not necessarily for consecutive years, and which extends for a period of more than three years. A "timeshare interest" is that interest purchased in a timeshare plan which grants the purchaser the right to use and occupy accommodations, facilities, or recreational sites, whether improved or unimproved, pursuant to a timeshare plan;
 - (54) The term "transfer" means every____
 - (A) the creation of a lien;
- (B) the retention of title as a security interest:
- (C) the foreclosure of a debtor's equity of redemption; or
- (D) each mode, direct or indirect, absolute or conditional, voluntary or involuntary, of disposing of or parting with...
 - <u>(i)</u> property; or with
- (ii) an interest in property, including retention of title as a security interest and foreclosure of the debtor's equity of redemption;
- (54A) the The term "uninsured State member bank" means a State member bank (as defined in section 3 of the Federal Deposit Insurance Act) the deposits of which are not insured by the Federal Deposit Insurance Corporation; and.
- (55) The term "United States", when used in a geographical sense, includes all locations where the judicial jurisdiction of the United States extends, including territories and possessions of the United States: [.]

§ 102. Rules of construction

In this title--

- (1) "after notice and a hearing", or a similar phrase-
- (A) means after such notice as is appropriate in the particular circumstances, and such opportunity for a hearing as is appropriate in the particular circumstances; but
 - (B) authorizes an act without an actual hearing if such notice is given properly and if-
 - (i) such a hearing is not requested timely by a party in interest; or
- (ii) there is insufficient time for a hearing to be commenced before such act must be done, and the court authorizes such act;
 - (2) "claim against the debtor" includes claim against property of the debtor;
 - (3) "includes" and "including" are not limiting;
- (4) "may not" is prohibitive, and not permissive;
- (5) "or" is not exclusive:
- (6) "order for relief" means entry of an order for relief;
- (7) the singular includes the plural:
- (8) a definition, contained in a section of this title that refers to another section of this title, does not, for the purpose of such reference, affect the meaning of a term used in such other section; and
- (9) "United States trustee" includes a designee of the United States trustee.

§ 103. Applicability of chapters

EXHIBIT D

- (B) with respect to an insured credit union (including an insured credit union for which the National Credit Union Administration has been appointed conservator or liquidating agent), the National Credit Union Administration;
- (C) with respect to any insured depository institution for which the Resolution Trust Corporation has been appointed conservator or receiver, the Resolution Trust Corporation; and
- (D) with respect to any insured depository institution for which the Federal Deposit Insurance Corporation has been appointed conservator or receiver, the Federal Deposit Insurance Corporation.
 - (22) the The term "financial institution" means-
- (A) means——(i) a Federal reserve bank, or an entity (domestic or foreign) that is a commercial or savings bank, industrial savings bank, savings and loan association, trust company, federally-insured credit union, or receiver, liquidating agent, or conservator for such entity and, when any such Federal reserve bank, receiver, liquidating agent, conservator; or entity is acting as agent or custodian for a customer in connection with a securities contract; (as defined in section 741-of this title, the) such customer; or
- (iiB) in connection with a securities contract, (as defined in section 741 of this title,) an investment company registered under the Investment Company Act of 1940; and 1940.
- (B) includes any person described in subparagraph (A) which operates, or operates as, a multilateral clearing-organization pursuant to section 409 of the Federal Deposit Insurance Corporation Improvement Act of 1991;22A) The term "financial participant" means--
- (A) an entity that, at the time it enters into a securities contract, commodity contract, swap agreement, repurchase agreement, or forward contract, or at the time of the date of the filing of the petition, has one or more agreements or transactions described in paragraph (1), (2), (3), (4), (5), or (6) of section 561(a) with the debtor or any other entity (other than an affiliate) of a total gross dollar value of not less than \$1,000,000,000 in notional or actual principal amount outstanding on any day during the previous 15-month period, or has gross mark-to-market positions of not less than \$100,000,000 (aggregated across counterparties) in one or more such agreements or transactions with the debtor or any other entity (other than an affiliate) on any day during the previous 15-month period; or
- (B) a clearing organization (as defined in section 402 of the Federal Deposit Insurance Corporation Improvement Act of 1991).
- (23) The term "foreign proceeding" means proceeding, whether collective judicial or administrative and whether or not under bankruptey law, in a foreign country in which the debtor's domicile, residence, principal place of business, or principal assets were located at the commencement of such proceeding, for the purpose of liquidating an estate, adjusting debts by composition, extension, or discharge, or effecting aproceeding in a foreign country, including an interim proceeding, under a law relating to insolvency or adjustment of debt in which proceeding the assets and affairs of the debtor are subject to control or supervision by a foreign court, for the purpose of reorganization; or liquidation.
- (24) The term "foreign representative" means duly selected trustee, administrator, or other representative of an estate in a foreign proceeding; a person or body, including a person or body appointed on an interim basis, authorized in a foreign proceeding to administer the reorganization or the liquidation of the debtor's assets or affairs or to act as a representative of such foreign proceeding.
- (25) The term "forward contract" means --
- (A) a contract (other than a commodity contract) for the purchase, sale, or transfer of a commodity, as defined in section 761(8) of this title, or any similar good, article, service, right, or interest which is presently or in the future becomes the subject of dealing in the forward contract trade, or product or byproduct thereof, with a maturity date more than two days after the date the contract is entered into, including, but not limited to, a repurchase transaction, reverse repurchase transaction, consignment, lease, swap, hedge transaction, deposit, loan, option, allocated transaction, or any other similar agreement
- (B) any combination thereof or option thereon; of agreements or transactions referred to in subparagraphs (A) and (C):
- (C) any option to enter into an agreement or transaction referred to in subparagraph (A) or (B);
- (D) a master agreement that provides for an agreement or transaction referred to in subparagraph (A), (B), or (C), together with all supplements to any such master agreement, without regard to whether such master agreement provides for an agreement or transaction that is not a forward contract under this paragraph, except that such master agreement shall be considered to be a forward contract under this paragraph only with respect to each agreement or transaction under such master agreement that is referred to in subparagraph (A), (B), or (C); or

- (E) any security agreement or arrangement, or other credit enhancement related to any agreement or transaction referred to in subparagraph (A), (B), (C), or (D), including any guarantee or reimbursement obligation by or to a forward contract merchant or financial participant in connection with any agreement or transaction referred to in any such subparagraph, but not to exceed the damages in connection with any such agreement or transaction, measured in accordance with section 562.
- (26) The term "forward contract merchant" means a person whose Federal reserve bank, or an entity the business of which consists in whole or in part of entering into forward contracts as or with merchants in a commodity, (as defined in section 761(8) of this title,) or any similar good, article, service, right, or interest which is presently or in the future becomes the subject of dealing in the forward contract trade;
- (27) The term "governmental unit" means United States; State; Commonwealth; District; Territory; municipality; foreign state; department, agency, or instrumentality of the United States (but not a United States trustee while serving as a trustee in a case under this title), a State, a Commonwealth, a District, a Territory, a municipality, or a foreign state; or other foreign or domestic government.
 - (27A) The term "health care business"--
- (A) means any public or private entity (without regard to whether that entity is organized for profit or not for profit) that is primarily engaged in offering to the general public facilities and services for--
 - (i) the diagnosis or treatment of injury, deformity, or disease; and
 - (ii) surgical, drug treatment, psychiatric, or obstetric care; and
 - (B) includes--
 - (i) any--
 - (I) general or specialized hospital;
 - (II) ancillary ambulatory, emergency, or surgical treatment facility;
 - (III) hospice:
 - (IV) home health agency; and
 - (V) other health care institution that is similar to an entity referred to in subclause (I), (II), (III), or (IV);

<u>and</u>

- (ii) any long-term care facility, including any--
 - (I) skilled nursing facility;
- (II) intermediate care facility;
- (III) assisted living facility:
- (IV) home for the aged;
- (V) domiciliary care facility; and
- (VI) health care institution that is related to a facility referred to in subclause (I), (II), (III), (IV), or (V), if that institution is primarily engaged in offering room, board, laundry, or personal assistance with activities of daily living and incidentals to activities of daily living.
 - (27B) The term "incidental property" means, with respect to a debtor's principal residence-
- (A) property commonly conveyed with a principal residence in the area where the real property is located;
- (B) all easements, rights, appurtenances, fixtures, rents, royalties, mineral rights, oil or gas rights or profits, water rights, escrow funds, or insurance proceeds; and
 - (C) all replacements or additions.
- (28) The term "indenture" means mortgage, deed of trust, or indenture, under which there is outstanding a security, other than a voting-trust certificate, constituting a claim against the debtor, a claim secured by a lien on any of the debtor's property, or an equity security of the debtor;
 - (29) The term "indenture trustee" means trustee under an indenture:
- (30) The term "individual with regular income" means individual whose income is sufficiently stable and regular to enable such individual to make payments under a plan under chapter 13 of this title, other than a stockbroker or a commodity broker.
 - (31) The term "insider" includes--
 - (A) if the debtor is an individual--
 - (i) relative of the debtor or of a general partner of the debtor:
 - (ii) partnership in which the debtor is a general partner;
 - (iii) general partner of the debtor; or
 - (iv) corporation of which the debtor is a director, officer, or person in control;
 - (B) if the debtor is a corporation-

EXHIBIT E

- (3) For the purposes of this section, a transfer is not made until the debtor has acquired rights in the property transferred.
- (f) For the purposes of this section, the debtor is presumed to have been insolvent on and during the 90 days immediately preceding the date of the filing of the petition.
- (g) For the purposes of this section, the trustee has the burden of proving the avoidability of a transfer under subsection (b) of this section, and the creditor or party in interest against whom recovery or avoidance is sought has the burden of proving the nonavoidability of a transfer under subsection (c) of this section.
- (h) The trustee may not avoid a transfer if such transfer was made as a part of an alternative repayment schedule between the debtor and any creditor of the debtor created by an approved nonprofit budgeting and credit counseling agency.
- (i) If the trustee avoids under subsection (b) a transfer made between 90 days and 1 year before the date of the filing of the petition, by the debtor to an entity that is not an insider for the benefit of a creditor that is an insider, such transfer shall be considered to be avoided under this section only with respect to the creditor that is an insider.

§ 548. Fraudulent transfers and obligations

- (a) (1) The trustee may avoid any transfer (including any transfer to or for the benefit of an insider under an employment contract) of an interest of the debtor in property, or any obligation (including any obligation to or for the benefit of an insider under an employment contract) incurred by the debtor, that was made or incurred on or within one year2 years before the date of the filing of the petition, if the debtor voluntarily or involuntarily-
- (A) made such transfer or incurred such obligation with actual intent to hinder, delay, or defraud any entity to which the debtor was or became, on or after the date that such transfer was made or such obligation was incurred, indebted; or
 - (B) (i) received less than a reasonably equivalent value in exchange for such transfer or obligation; and
- (ii) (I) was insolvent on the date that such transfer was made or such obligation was incurred, or became insolvent as a result of such transfer or obligation;
- (II) was engaged in business or a transaction, or was about to engage in business or a transaction, for which any property remaining with the debtor was an unreasonably small capital; or
- (III) intended to incur, or believed that the debtor would incur, debts that would be beyond the debtor's ability to pay as such debts matured; or
- (IV) made such transfer to or for the benefit of an insider, or incurred such obligation to or for the benefit of an insider, under an employment contract and not in the ordinary course of business.
- (2) A transfer of a charitable contribution to a qualified religious or charitable entity or organization shall not be considered to be a transfer covered under paragraph (1)(B) in any case in which--
- (A) the amount of that contribution does not exceed 15 percent of the gross annual income of the debtor for the year in which the transfer of the contribution is made; or
- (B) the contribution made by a debtor exceeded the percentage amount of gross annual income specified in subparagraph (A), if the transfer was consistent with the practices of the debtor in making charitable contributions.
- (b) The trustee of a partnership debtor may avoid any transfer of an interest of the debtor in property, or any obligation incurred by the debtor, that was made or incurred on or within one year 2 years before the date of the filing of the petition, to a general partner in the debtor, if the debtor was insolvent on the date such transfer was made or such obligation was incurred, or became insolvent as a result of such transfer or obligation.
- (c) Except to the extent that a transfer or obligation voidable under this section is voidable under section 544, 545, or 547 of this title, a transferee or obligee of such a transfer or obligation that takes for value and in good faith has a lien on or may retain any interest transferred or may enforce any obligation incurred, as the case may be, to the extent that such transferee or obligee gave value to the debtor in exchange for such transfer or obligation.

- (d) (1) For the purposes of this section, a transfer is made when such transfer is so perfected that a bona fide purchaser from the debtor against whom applicable law permits such transfer to be perfected cannot acquire an interest in the property transferred that is superior to the interest in such property of the transferee, but if such transfer is not so perfected before the commencement of the case, such transfer is made immediately before the date of the filing of the petition.
 - (2) In this section--
- (A) "value" means property, or satisfaction or securing of a present or antecedent debt of the debtor, but does not include an unperformed promise to furnish support to the debtor or to a relative of the debtor;
- (B) a commodity broker, forward contract merchant, stockbroker, financial institution, <u>financial participant</u> or securities clearing agency that receives a margin payment, as defined in section 101, 741, or 761 of this title, or settlement payment, as defined in section 101 or 741 of this title, takes for value to the extent of such payment;
- (C) a repo participant or financial participant that receives a margin payment, as defined in section 741 or 761 of this title, or settlement payment, as defined in section 741 of this title, in connection with a repurchase agreement, takes for value to the extent of such payment; and
- (D) a swap participant or financial participant that receives a transfer in connection with a swap agreement takes for value to the extent of such transfer; and
- (E) a master netting agreement participant that receives a transfer in connection with a master netting agreement or any individual contract covered thereby takes for value to the extent of such transfer, except that, with respect to a transfer under any individual contract covered thereby, to the extent that such master netting agreement participant otherwise did not take (or is otherwise not deemed to have taken) such transfer for value.
- (3) In this section, the term "charitable contribution" means a charitable contribution, as that term is defined in section 170(c) of the Internal Revenue Code of 1986, if that contribution--
 - (A) is made by a natural person; and
 - (B) consists of--
- (i) a financial instrument (as that term is defined in section 731(c)(2)(C) of the Internal Revenue Code of 1986); or
 - (ii) cash.
- (4) In this section, the term "qualified religious or charitable entity or organization" means-
 - (A) an entity described in section 170(c)(1) of the Internal Revenue Code of 1986; or
 - (B) an entity or organization described in section 170(c)(2) of the Internal Revenue Code of 1986.
- (e)(1) In addition to any transfer that the trustee may otherwise avoid, the trustee may avoid any transfer of an interest of the debtor in property that was made on or within 10 years before the date of the filing of the petition if—
- (A) such transfer was made to a self-settled trust or similar device;
- (B) such transfer was by the debtor:
- (C) the debtor is a beneficiary of such trust or similar device; and
- (D) the debtor made such transfer with actual intent to hinder, delay, or defraud any entity to which the debtor was or became, on or after the date that such transfer was made, indebted.
- (2) For the purposes of this subsection, a transfer includes a transfer made in anticipation of any money judgment, settlement, civil penalty, equitable order, or criminal fine incurred by, or which the debtor believed would be incurred by—
- (A) any violation of the securities laws (as defined in section 3(a)(47) of the Securities Exchange Act of 1934 (15 U.S.C. 78c(a)(47))), any State securities laws, or any regulation or order issued under Federal securities laws of State securities laws; or
- (B) fraud, deceit, or manipulation in a fiduciary capacity or in connection with the purchase or sale of any security registered under section 12 of 15(d) of the Securities Exchange Act of 1934 (15 U.S.C. 781 and 78o(d)) or under section 6 of the Securities Act of 1933 (15 U.S.C. 77f).

§ 549. Postpetition transactions

- (a) Except as provided in subsection (b) or (c) of this section, the trustee may avoid a transfer of property of the estate--
- (1) that occurs after the commencement of the case; and
- (2) (A) that is authorized only under section 303(f) or 542(c) of this title; or